

# Effects of climate, soil, and forest structure on specific leaf area of understorey foundation plants

Sashank Khaniya

Master's thesis • 30 credits Swedish University of Agricultural Sciences, SLU Southern Swedish Forest Research Centre Sustainable Forest and Nature Management (SUFONAMA) Alnarp 2023

# Effects of climate, soil, and forest structure on specific leaf area of understorey foundation plants

Sashank Khaniya

Supervisor:	Per-Ola Hedwall, Swedish University of Agricultural Sciences,
	Southern Swedish Forest Research Centre
Assistant supervisor:	Joan Díaz Calafat, Swedish University of Agricultural Sciences,
	Southern Swedish Forest Research Centre
Examiner:	Jörg Brunet, Swedish University of Agricultural Sciences, Southern
	Swedish Forest Research Centre

Credits:	30 credits				
Level:	Second cycle, A2E				
Course title:	Master's Thesis in Forest Science				
Course code:	EX0984				
Programme/education:	Sustainable Forest and Nature Management (SUFONAMA)				
Course coordinating dept:	Southern Swedish Forest Research Centre				
Place of publication:	Alnarp				
Year of publication:	2023				
Copyright:	All featured images are used with permission from the copyright owner.				
Keywords:	Specific leaf area, understorey vegetation, foundation species, forest structure, climate change				

Swedish University of Agricultural Sciences Faculty of Forest Sciences Southern Swedish Forest Research Centre

#### Abstract

Forest overstorey's interaction with understorey vegetation is a crucial aspect that impacts the biodiversity and resilience of forest ecosystems. Particularly, the influence of the overstorey on the specific leaf area (SLA) in understorey plants remains poorly understood. This is due to limited research within particular climatic conditions and a narrow focus on limited biogeographical areas, which leaves significant knowledge gaps, especially in boreal regions. This study aims to address these gaps by exploring how the SLA of understorey plants responds to variations in forest structure, soil characteristics, and climate across diverse Swedish forests. To achieve this aim, this study embodied three distinct study designs, namely a large-scale study, a mixed forest study, and a transplant experiment. Cowberry and bilberry were chosen for the large-scale and mixed forest studies, while wild strawberry and goldenrod were selected for the transplant experiment. Leaf samples of these species were used for this study. Cowberry being evergreen, the leaf samples were collected from old shoots and new shoots, which are termed as "cowberry old growth" and "cowberry new growth", respectively. The large-scale study consisted of three sites covering latitudes of 57° N to 64° N, each with 15-16 stands dominated by Scots pine or Norway spruce. The mixed forest study comprised a stand mainly with Norway spruce and birch where 40 plots were strategically placed across gradients of varying forest density and overstorey species composition. For the transplant experiment, understorey plants were planted in pairs with different climatic origins, at three sites in Sweden. Plants were planted in ten plots per site, established at diverse tree species composition and forest density values. Forest density had a significant positive effect on the SLA of understorey plants. The percentage of pine had a significant negative effect on SLA in the large-scale study. The percentage of broadleaves had a significant negative effect on SLA in mixed forest study and transplant experiment. Soil C-N ratio did not have significant effect on the SLA. Soil moisture availability had no significant impact on the SLA of the understorey plants except for the cowberry new growth in large-scale study and bilberry in mixed forest study. The interaction between forest density and the percentage of pine only significantly affected the cowberry new growth in the large-scale study. Furthermore, the SLA of wild strawberry and goldenrod was not affected by climatic origin, and the interaction between forest density and origin was also not significant. These findings help to predict how the SLA of understorey plants changes related to forest structure, as well as to changes induced by climate warming, while at the same time enriching our insight into plant-environment interactions.

*Keywords*: Specific leaf area, understorey vegetation, foundation species, forest structure, climate change

# Table of contents

List	of tables	6
List	of figures	7
Abbr	reviations	8
1.	Introduction	9
1.1	Objectives	12
2.	Methodology	13
2.1	Study area and Study design	14
	2.1.1 Large-scale study	14
	2.1.2 Mixed forest study	14
	2.1.3 Transplant experiment	15
2.2	Data collection	16
	2.2.1 Collection of samples from understorey plants	16
	2.2.2 Collection of overstorey data	16
	2.2.3 Collection of soil data	16
2.3	Data preparation	17
2.4	Data analysis	18
3.	Results	20
3.1	Large-scale study	20
	3.1.1 Model prediction for large-scale study	22
3.2	Mixed forest study	23
	3.2.1 Model prediction for mixed forest study	26
3.3	Transplant experiment	27
	3.3.1 Model prediction for transplant experiment	30
4.	Discussion	31
4.1	Effect of overstorey structure and composition	32
4.2	Effect of soil moisture and carbon-nitrogen ratio	33
4.3	Effect of warming	35
4.4	Implications for forest management	36
5.	Conclusion and Recommendations	38

References	40
Popular science summary	49
Acknowledgements	51

# List of tables

# List of figures

Figure 1.	Map of Sweden showing study locations. a) the large-scale study, b) the mix	ed
	forest study, c) the transplant experiment (Triangle shapes represent co	old
	regions whereas the solid circles represent the warm regions. Transplant w	as
	done from Tärnaby, Idre, and Tomtabacken to Vindeln, Siljan, and Viva	rp,
	respectively.)	13
Figure 2.	Sampling design for the large-scale study	14

- Figure 3. Procedures for SLA calculation.....17

# Abbreviations

AIC	Akaike Information Criterion			
BA	Total Basal Area			
C-N	Carbon Nitrogen			
DBH	Diameter at Breast Height			
GLMMs	Generalized Linear Mixed Models			
SLA	Specific Leaf Area			
SLU	Swedish University of Agricultural Sciences			
SM	Soil Moisture			
% pine	Percentage of pine			
% broadleaves	Percentage of broadleaves			

### 1. Introduction

Forests are complex ecosystems that comprise different biotic and abiotic components, which interact among themselves and contribute to ecosystem functioning. While trees are the dominant structures in a forest ecosystem, it also hosts understorey vegetation which includes seedlings, herbs, shrubs, mosses, and lichens (Deng et al. 2023). Several studies have reported the role of the understorey in the functioning of forest ecosystems such as nutrient cycling, water balance, and carbon sequestration (Balandier et al. 2022a; b; Deng et al. 2023; Duan et al. 2023). In addition, understorey plants support biodiversity by providing food and habitat to animals, insects, and soil arthropods (Balandier et al. 2022b; Xiao et al. 2022; Zhang et al. 2022; Deng et al. 2023). Understorey plants are also found to increase bird diversity in the Mediterranean region (Dagan & Izhaki 2019).

The overstorey in northern forests can be composed of deciduous trees or evergreen trees, often in mixtures. Likewise, the overstorey trees also may vary in age and height. Such factors combinedly shape the overstorey forest structure. The heterogeneity in overstorey structure controls the amount of light entering the forest and thus microclimate buffering (Balandier et al. 2022b; Christiansen et al. 2022; Deng et al. 2023). Natural disturbance is considered an important factor that maintains the structural heterogeneity of the overstorey (Kuuluvainen et al. 2021). Naturally, forest disturbance is caused by biotic agents like insects or by events of windthrow and fire, and thus helps in maintaining biodiversity in the forest through species reassembly (Kuuluvainen et al. 2021). However, climate change has increased the frequency and severity of these natural disturbances (Seidl et al. 2014, 2017; Romeiro et al. 2022) and have resulted in adverse effects on the forest ecosystem. For instance, the tree canopy mortality rate has increased on average by 0.79% per year in Europe from 1984 to 2016 (Senf et al. 2018), which is attributed to climate change induced forest disturbances. This canopy mortality was found to increase by 0.41% for a 1°C rise in mean annual temperature (Senf et al. 2018). Moreover, the damage to timber volume from disturbances such as wind, bark beetle, and fire is predicted to increase by ca 230%, 760%, and 310% respectively in 2011-2030 as compared to 1971-1980 (Seidl et al. 2014). Similarly, a significant positive trend in the impacts of natural disturbance in Europe was documented in a study by Patacca et al. (2023). Such evidence

indicates that the severity of forest disturbances will increase leading to increased tree mortality, which will likely promote canopy openness and decrease tree density in the forest (Kumar et al. 2021). A study by Thom et al. (2020) shows that the amount of light reaching the understorey can increase by 227% for distributed disturbances to 387% for aggregated disturbances relative to the undisturbed area. Also, a 10% increase in light in the understorey was found to increase microclimatic temperature variation by 0.11°C (Thom et al. 2020). Another factor that affects overstorey structure is anthropogenic disturbances such as logging, overgrazing, and deforestation (Mcdowell et al. 2020). Silvicultural activities such as thinning, pruning, felling, and weeding also change forest structure which further effects light and water interception, evapotranspiration, and ultimately altering microclimate inside the forest which affects understorey plants (Balandier et al. 2022b; Christiansen et al. 2022; Deng et al. 2023).

Understorey plants respond to such changes in light and microclimate by altering their functional traits (Funk et al. 2017; Blondeel et al. 2020; Kemppinen & Niittynen 2022). Plant functional traits - characteristics of plant species that influence their ecological function and interactions with the environment, other species, and different trophic levels, are increasingly being used as a reliable and effective approach to investigate a direct link between individual plants and ecosystem-level functions (Pérez-Harguindeguy et al. 2016). Furthermore, functional traits can help explain the response of plants to past and future climates (Heilmeier 2019).

Among such traits, specific leaf area (SLA) is a commonly used trait for ecological studies. SLA is defined as "the one-sided area of a leaf divided by its oven-dry mass, expressed in  $m^2/kg$  or in  $mm^2/mg$ " (Cornelissen et al. 2003). SLA is a valuable indicator of plant performance and ecological strategies, as it reflects the balance between growth and resource acquisition rate levels (Pérez-Harguindeguy et al. 2016). SLA is also crucial in influencing the productivity of plants, as variations in SLA indicate changes in leaf structure and therefore their nutritional composition. SLA was, for example, found to be the best predictor of tree species response to climate change in mixed sub-tropical forests (Schlickmann et al. 2020). Species growing in areas with abundant resources tend to have higher SLA compared to species growing in resource-deficit areas (Cornelissen et al. 2003; Pérez-Harguindeguy et al. 2016). Therefore, we can consider SLA to be a proxy for the costliness of leaf production. Leaves with higher SLA (lower thickness and/or density) are more efficient in utilizing low amounts of diffuse radiation, and they have greater relative growth rates under favourable conditions. On the other hand, leaves with lower SLA (greater thickness and/or density) are more efficient in utilizing high levels of radiation and are more tolerant to water and nutrient deficiencies (Burns 2004). Such plasticity of SLA can increase the resilience of plants to disturbance and changing

environmental conditions (Römermann et al. 2011). Additionally, SLA influences the expansion and growth of plant canopies by affecting the overall leaf area per plant, thereby impacting light interception and utilization efficiency (Poorter et al. 2019).

Plant species characterized by high SLA demonstrate increased nutrient absorption and photosynthetic capacity, contributing significantly to the primary productivity of the community (Wright et al. 2001; Chelli et al. 2019; Anderson et al. 2020), and thus reflecting their importance within the community. This also implies greater availability of food for herbivorous organisms. A higher SLA is also linked to increased susceptibility to herbivory due to the higher palatability of the tissue (Funk et al. 2017). Furthermore, variations in SLA influence the changes in vegetation dynamics in an ecosystem ultimately affecting ecosystem processes, including energy flow and nutrient cycling. A review by Shome et al. (2023) provides evidence for a positive association of SLA with several ecosystem services including water regulation, biomass production, litter decomposition, nutrient cycling, and aesthetic appeal. The same review shows a strong negative association of SLA with phytoremediation, a process by which plants naturally absorbs, stores, and transform harmful chemicals from soil, water, and air (Shome et al. 2023). Additionally, previous studies have documented a strong positive correlation of SLA with aboveground net primary productivity (Garnier et al. 2004), and aboveground biomass (Ali et al. 2017). Moreover, SLA plays a role in determining the amount of new leaf area produced per unit of biomass (Niklas et al. 2007), and it directly affects the decomposition process by influencing the relative exposure of leaf litter surfaces to abiotic factors such as UV light, soil moisture, and soil microbes (Liu et al. 2018). High SLA values are also associated with higher litter decomposition rates (Garnier et al. 2004; Mokany et al. 2008), implying that species with thinner leaves experience a faster breakdown of organic matter. SLA has been found to have a negative relationship with aboveground carbon stocks of trees, and this association increases with increasing spatial scale (Bu et al. 2019). Additionally, SLA is negatively related to soil C-N (carbon to nitrogen) on a global scale (Garnier et al. 2004; Ordoñez et al. 2009), and it is also negatively correlated with total soil carbon and total soil nitrogen while being positively correlated with the rate of litter mass loss (Garnier et al. 2004). Such correlation of SLA to diverse factors which in turn shapes ecosystem functioning highlights the significance of the use of SLA in ecological studies.

The factors that impact SLA may vary based on the plant species and environmental conditions but typically include factors such as light, nutrients, water, temperature, and disturbances (Poorter et al. 2009; Gong & Gao 2019; Blondeel et al. 2020; De Pauw et al. 2022). Previous studies on the effect of such factors on SLA have been limited to homogenous climatic conditions (Wasof et al. 2013) or based on limited biogeographic regions (Chelli et al. 2019). Yet, such studies in boreal regions are scarce. A comprehensive understanding of such factors can help to predict how individual plant species and ecosystems may respond to environmental changes and provide valuable insights into plant-environment interactions. Moreover, the relationship between the SLA of understorey plants and forest structure, along with the combined effect of climate is yet to be explored. Quantification of such relationship is crucial for forest management decisions and to study the effects of climate change.

This study endeavours to bridge the knowledge gap on how the SLA of understorey plants responds to varying forest structures, soil, and climate. The outcomes of this study can be insightful for policymakers in framing plans for reducing the effects of climate change and land-use impacts and adapting to them in preserving biodiversity and improving ecosystem services. Also, it will assist as a reference for national and international organizations involved in conserving biodiversity and lowering the effect of climate change on understorey vegetation in similar systems.

### 1.1 Objectives

The general objective of this study was to identify key drivers of SLA of understorey plants.

The specific objectives of this study are as follows:

Objective 1: To quantify the influence of forest structure on the SLA of understorey plants.

Research question: How does variation in forest structure influence the SLA of understorey plants?

Objective 2: To investigate the relationship between soil variables and the SLA of understorey plants.

Research question: How do soil nutrient and soil moisture availability affect the SLA of understorey plants?

Objective 3: To investigate the effect of climate change on understorey plants. Research question: How does warming affect the SLA of understorey plants?

# 2. Methodology



Figure 1. Map of Sweden showing study locations. a) the large-scale study, b) the mixed forest study, c) the transplant experiment (Triangle shapes represent cold regions whereas the solid circles represent the warm regions. Transplant was done from Tärnaby, Idre, and Tomtabacken to Vindeln, Siljan, and Vivarp, respectively.)

This study was based on available datasets from three different studies, here called the large-scale study, the mixed forest study, and the transplant experiment. All the sites used in the study were located in Sweden (Figure 1). The understorey plants used for the large-scale study and mixed forest study were cowberry (*Vaccinium vitis-idaea* L.), and bilberry (*Vaccinium myrtillus* L.). Wild strawberry (*Fragaria vesca* L.), and goldenrod (*Solidago virgaurea* L.) were selected for the transplant experiment.

Cowberry, bilberry, wild strawberry, and goldenrod are common understorey species occurring all over Sweden (Sjörs 1999; Nilsson & Wardle 2005). Cowberry and bilberry are woody dwarf shrubs, whereas wild strawberry and goldenrod are herbaceous. Regarding the dwarf shrubs, cowberry is evergreen whereas bilberry sheds its leaves yearly in autumn (Ritchie 1955, 1956). Moreover, cowberry and bilberry often dominate the ground vegetation, provide fruits, have significant positive impacts on soil microbial activities and litter decomposition, influence the regeneration of tree seedlings, and play a significant role in the functioning of the forest ecosystem (Nilsson & Wardle 2005). Such characteristics qualify cowberry and bilberry as foundation species, which can be defined as "a species (or group of functionally similar taxa) that dominates an assemblage numerically and in overall size (usually mass), determines the diversity of associated taxa through non-trophic interactions, and modulates

fluxes of nutrients and energy at multiple control points in the ecosystem it defines" (Ellison 2019).

### 2.1 Study area and Study design

#### 2.1.1 Large-scale study



Figure 2. Sampling design for the large-scale study

The study design and data collection for the large-scale study were done in August and early September 2020. The conceptual figure for study design is shown in Figure 2. Three sites (Asa, Siljansfors, and Vindeln; Figure 1a), were selected, spanning a latitudinal gradient from 57° N to 64° N to include a wide environmental and climatic variation. Asa and Siljansfors had 16 forest stands each whereas Vindeln had 15 stands. The stands were selected based on species composition, age, and stand fertility. This data was already available at each SLU field station. The stands were divided into four categories based on age: early (6-15 years), young (16-48 years), middle (49-75 years), and old (>76 years old). The stands were either dominated by Scots pine (*Pinus sylvestris* L.; hereafter pine) or Norway spruce (Picea abies (L.) H. Karst.; hereafter Norway spruce). Three circular plots (which acted as replicates) of 5 m radius (in the young stand) or 8 m (for the other tree age categories) were placed in each stand to measure tree data and soil data. A 4 m<sup>2</sup> square sub-plot was located in the centre of each plot which further consisted of sixteen 0.25 m<sup>2</sup> squares. Three sub-plots were selected randomly in each plot, where all understorey plant measurements and sampling took place.

#### 2.1.2 Mixed forest study

The mixed forest study was based on a single stand in Vindeln (Figure 1b). The stand consisted of a mixed forest with mainly Norway spruce and birch (*Betula pendula* Roth and *B. pubescens* Ehrh.; hereafter birch) and other broadleaves in

small amounts. Within this stand, 40 circular plots with a 10 m radius were selected. These plots were located across a double gradient of forest density and species composition so that they captured the maximum variability of forest density and tree species composition. In each plot, a sub-plot of 2  $m^2$  was established, from which understorey plant biomass was collected. Samples were collected in July 2020.

#### 2.1.3 Transplant experiment

In the transplant experiment, wild strawberry and goldenrod were collected from six sites in Sweden, organized into three pairs (Figure 1c). Each pair consisted of one site with a colder climate where the mean annual temperature was on average 2°C lower (Tomtabacken, Idre, and Tärnaby), and one site with a warmer climate (Siljan, Vindeln, and Vivarp). In 2020, plants original to the cold provenances were translocated to their relatively warmer sites (hereafter called *cold origin*). There, local plants were also repotted to standardize soil and stress conditions (hereafter called *local origin*).

The collection of understorey plants was done at 2-4 places, which were placed at least 1 km apart from each other, within forest stands at each location (both cold and warm sites). For wild strawberry, a distance of at least 7 m between individuals was considered during collection to avoid collecting ramets from the same clone. For goldenrod, individuals with mature seed were collected without taking distance between individuals into consideration. The collection was done during summer of 2020. Wild strawberry was then planted in container trays and kept in a garden at each warm site until replanting in the bigger pots, and finally placed in the forest in spring 2021. The goldenrod seeds were stored over winter and germinated. They were planted in container trays in spring and summer 2021 and were placed in the forest in August 2021. In each warmer site, 10 circular plots of a 10 m radius were established within a single forest stand. Forest stands consisted mainly of Norway spruce and birch. Large variations in tree species composition and forest density were present within these plots, allowing for the capture of maximum variability in both factors. A 7.5 L pot was used, and four plant individuals were planted in each pot. The pots were inserted into the soil at the destination site and plot with a spade. A soil with medium pH- "Hasselfors Pjord" was used. All plots within a site had similar soil conditions and slight altitudinal variations. Additionally, the plots in Vivarp were fenced to protect them from big herbivores. Strawberries were sampled in August 2021 and 2022 whereas goldenrods were sampled in August 2022.

### 2.2 Data collection

#### 2.2.1 Collection of samples from understorey plants

The rhizomatous nature of understorey plants (bilberry and cowberry) limited the sampling of individual plants. Therefore, three clonal units per species (hereafter called ramets) that were free from damage were selected randomly in each subplot in the large-scale study. In the mixed forest study, six ramets were collected for bilberry and cowberry. The total number of terminal shoots per ramet was recorded, and shoots growing from the ramets were detached and air-dried. For cowberry, the leaf samples were collected from old shoots and new shoots, which are hereafter called "cowberry old growth" and "cowberry new growth", respectively. New shoots were separated from old shoots by comparing the colour of the stem and leaves to older biomass. Bilberry is deciduous and hence only newly grown leaves samples were collected. The leaves were selected randomly, ensuring that they represented the various leaf sizes present within each ramet and maintaining the proportional occurrence of these sizes. A total of 50 leaves were sampled from each bilberry ramet and 25 from each cowberry ramet and growth section (i.e., old, and new growth). In instances where only a smaller quantity of leaves was available, all of them were utilized.

In the case of the transplant experiment, one leaf from each pot of wild strawberry and goldenrod (of both cold and local origin) was collected in August 2021.

### 2.2.2 Collection of overstorey data

The species of all living trees taller than 1.3 m and their diameter at breast height (DBH) were recorded in each plot. The total basal area per plot (BA) and basal area contribution of each tree species were calculated. BA was used as a proxy for forest density in this study and was calculated in units of  $m^2/ha$ . Further, the percentage of basal area of broadleaves (% broadleaves; for the mixed and transplant experiments) or pine trees (% pine; for the large-scale experiment) per plot was calculated from the basal area contribution of each tree species. The age of measured trees at 1.3 m was determined using a core sample taken at breast height. The height of the tree with the largest DBH was measured in two plots per stand. If the largest tree had a damaged or crooked stem, the next biggest tree was selected.

### 2.2.3 Collection of soil data

Four soil samples within each circular plot and outside the 4  $m^2$  sub-plots for plant data collection were collected in the large-scale study and the mixed forest study. Litter was removed from the soil surface before sampling. Any rock and

other impediments were avoided during the collection of soil. A 10 cm deep soil sample including both humus and mineral soil was collected. Using a Theta Probe (Type ML1; Delta-T Devices, Cambridge, UK), soil moisture (SM) was measured and was expressed in percentage. In the mixed forest study, SM was measured taking three measurements per sub-plot, taken within 2 m from the sampled ramet. The three measurements were taken at different locations within the 4 m<sup>2</sup> sub-plot, trying to account for the sub-plot heterogeneity.

All soil samples from the same plot were pooled and dried at 60°C to remove most SM. The soil sample was then analyzed for total carbon and nitrogen content with an Avio 200 ICP spectrometer (PerkinElmer, Waltham, Massachusetts, USA), from which the C-N ratio was calculated.

### 2.3 Data preparation



Figure 3. Procedures for SLA calculation

SLA (in mm<sup>2</sup>/mg) for each leaf sample was determined following standard procedures (Cornelissen et al. 2003; Pérez-Harguindeguy et al. 2016). The procedure involves leaf sample scanning, drying, and measuring the dry weight of leaves as shown in Figure 3. Leaf samples were scanned using a flatbed scanner. The leaf area of the scanned leaves was calculated by importing scanned images of leaf samples into R version 4.1.3 (R Core Team 2022) and utilizing the 'LeafArea' package, which enables leaf area calculation through batch-processing of several images in the ImageJ software (Katabuchi 2015). After scanning leaf samples, they were dried in an oven at 60°C for 48hrs. SLA was calculated for each leaf sample using Equation 1.

$$SLA = \frac{One \ sided \ area \ of \ leaf}{Oven \ dry \ weight \ of \ the \ leaf}$$
(1)

The available explanatory variables were checked for collinearity by making scatter plots in R and checking their correlation with Spearman's correlation coefficients. Variables with a Spearman's correlation coefficient above 0.7 were removed from further analysis to ensure the inclusion of independent variables with minimal multicollinearity. Finally, the selected explanatory variable for each study is presented in Table 1.

Explanatory variables				
Continuous	Categorical			
Total Basal Area (BA)	Stand			
Percentage of pine (% pine)				
C-N ratio				
Soil Moisture (SM)				
Total Basal Area (BA)				
Percentage of broadleaves (%				
broadleaves)				
C-N ratio				
Soil Moisture (SM)				
Total Basal Area (BA)	Site			
Percentage of broadleaves (%	Origin (Cold, Warm)			
broadleaves)				
	Explanatory vaContinuousTotal Basal Area (BA)Percentage of pine (% pine)C-N ratioSoil Moisture (SM)Total Basal Area (BA)Percentage of broadleaves (%broadleaves)C-N ratioSoil Moisture (SM)Total Basal Area (BA)Percentage of broadleaves (%broadleaves)C-N ratioSoil Moisture (SM)Total Basal Area (BA)Percentage of broadleaves (%broadleaves)Contal Basal Area (BA)Contal Basal Area (BA)Soil Moisture (SM)Total Basal Area (BA)Percentage of broadleaves (%broadleaves)			

Table 1. Variables selected for model generation and their abbreviations used in the text.

### 2.4 Data analysis

All data analyses were performed in R version 4.1.3 (R Core Team 2022). All the continuous explanatory variables were standardized before further analysis by subtracting their respective mean and dividing by their standard deviation. This standardization of variables ensures that all explanatory variables are on the same scale and magnitudes of coefficients can be directly compared. The coefficient values of explanatory variables mirror the change in the dependent variable (i.e., SLA in this study) linked with a one standard deviation increase or decrease in the standardized explanatory variable. Larger coefficient values of the explanatory variable indicate a more robust association with SLA. Generalized linear mixed models (GLMMs) were fitted using the glmmTMB function of the 'glmmTMB' package (Brooks et al. 2017). The 'glmmTMB' package was chosen for its ability to handle a wide range of response distributions which enables the selection of the best distribution and link function depending on the characteristics of the response variable and its efficient optimization techniques (adaptive Gaussian quadrature or Laplace approximation) (Brooks et al. 2017). In addition, the package allowed for the inclusion of random effects to account for within-group correlation, which is necessary when the data is clustered or hierarchical. The data used in this study had a hierarchical structure (i.e., multiple plots were present within a site). Gaussian distribution was chosen for the fitted models. Stand (in the large-scale study) and Site (in the transplant experiment) were included as random intercept effects in the fitted models to account for grouping or nesting (i.e., observations in the same stand or site are likely to be similar as compared to observations between different stands and site). So, by including these variables as random intercept effects, the difference in SLA between the stand or site is accounted for. The fitted models are presented in Table 2.

Table 2. Table showing candidate models for the various studies. Note that Stand and Site were always used as random effects.

Study	Fitted models
Large-scale study	<b>Model 1</b> : $SLA = BA \times \%$ pine + C-N ratio + SM + Stand
	<b>Model 2</b> : SLA = BA + % pine + C-N ratio + SM + Stand
Mixed forest study	<b>Model 1</b> : $SLA = BA \times \%$ broadleaves + C-N ratio+ SM
	<b>Model 2</b> : SLA = BA + % broadleaves + C-N ratio+ SM
Transplant experiment	<b>Model 1</b> : $SLA = BA \times \%$ broadleaves + Site
	<b>Model 2</b> : SLA = BA + % broadleaves + Site
	<b>Model 3</b> : $SLA = BA \times Origin + Site$
	<b>Model 4</b> : SLA = BA + Origin + Site

Finally, SLA was predicted for low, mean, and high values of % pine for largescale study and low, mean, and high values of % broadleaves for mixed forest and transplant experiments. Akaike Information Criterion (AIC) was used to select the best model, lower AIC indicating a better fit of the model. Geospatial package (Dunnington 2022) was used to create maps of the study area, and the *ggplot* function of the 'ggplot2' package (Wickham 2016) was used to create all the graphs.

# 3. Results

### 3.1 Large-scale study

The results demonstrate Model 2 (AIC = 820.6) shows slightly better fit when compared with Model 1 (AIC = 821.5) for bilberry. Both Model 1 (Z value = 3.56, p < 0.001) and Model 2 (Z value = 4.13, p < 0.001) show the effect of BA on SLA was significantly positive whereas both models showed that % pine had a significant negative effect on SLA (Z value = -2.12, p < 0.05 in Model 1; Z value = -2.04, p < 0.05 in Model 2) (Table 3). However, the C-N ratio (p = 0.311 in Model 1; p = 0.223 in Model 2) and SM (p = 0.791 in Model 1; p = 0.633 in Model 2) did not show significant effects on SLA in any of the models (Table 3). Furthermore, the interaction between BA and % pine was not significant (Z value = -1.04, p = 0.301) (Table 3).

Table 3. Summary of model output for bilberry in the large-scale study.

AIC = 821.5				
Variables	Estimate	Standard Error	Z value	<b>Pr</b> (>  <b>z</b>  )
Intercept	7.36	0.47	15.56	< 0.001
BA	1.68	0.47	3.56	< 0.001
% pine	-0.99	0.47	-2.12	0.034
C-N ratio	-0.62	0.62	-1.01	0.311
SM	-0.19	0.45	-0.27	0.791
$BA \times \%$ pine	-0.51	0.49	-1.04	0.301

*Model 1* :  $SLA = BA \times \%$  pine + C-N ratio + SM + Stand (as random intercept effect)  $\Delta IC = 821.5$ 

*Model 2*: *SLA* = *BA* + % *pine* + *C*-*N ratio* + *SM* + *Stand* (*as random intercept effect*) AIC = 820.6

Variables	Estimate	Standard Error	Z value	<b>Pr(&gt; z )</b>
Intercept	7.442	0.49	15.03	< 0.001
BA	1.84	0.45	4.13	< 0.001
% pine	-0.95	0.47	-2.04	0.041
C-N ratio	-0.74	0.61	-1.22	0.223
SM	-0.21	0.44	-0.49	0.633

For cowberry new growth, Model 1 (AIC = 432) had better fit than Model 2 (AIC = 442.5). Similar to the results of bilberry, both Model 1 (Z value = 7.83, p < 0.001) and Model 2 (Z value = 8.80, p < 0.001) showed that BA had a significant positive effect on the SLA of cowberry new growth whereas % pine had a significant negative effect on SLA for both Model 1 (Z value = -8.46, p < 0.001) and Model 2 (Z value = -7.48, p < 0.001) (Table 4). While the C-N ratio did not show a significant effect on SLA in any of the models, SM had a significant negative effect in Model 2 (Z value = -2.29, p < 0.05) (Table 4). Furthermore, the interaction between BA and % pine was significant in Model 1 (Z value = -3.51, p < 0.001), suggesting that the proportion of pine influences the effect of BA on SLA (Table 4).

Table 4. Summary of model output for cowberry new growth in the large-scale study.

<i>Model 1</i> : $SLA = BA \times \%$ pine + C-N ratio + SM + Stand (as random intercept effect)				
AIC = 432				
Variables	Estimate	Standard	Z value	<b>Pr(&gt; z )</b>
		Error		
Intercept	3.65	0.24	15.32	< 0.001
BA	1.26	0.16	7.83	< 0.001
% pine	-1.42	0.17	-8.46	< 0.001
C-N ratio	-0.11	0.20	-0.52	0.604
SM	-0.20	0.15	-1.36	0.174
$BA \times \%$ pine	-0.63	0.18	-3.51	< 0.001

*Model 2: SLA* = *BA* + % *pine* + *C*-*N ratio* + *SM* + *Stand* (*as random intercept effect*) AIC = 442.5

Variables	Estimate	Standard	Z value	<b>Pr(&gt; z )</b>
		Error		
Intercept	3.71	0.24	15.42	< 0.001
BA	1.42	0.16	8.80	< 0.001
% pine	-1.28	0.17	-7.48	< 0.001
C-N ratio	-0.22	0.21	-1.04	0.3
SM	-0.35	0.15	-2.29	0.021

For cowberry old growth, Model 2 (AIC = 434.1) had better fit than Model 1 (AIC = 435.3). Both Model 1 (Z value = 4.07, p < 0.001) and Model 2 (Z value = 4.39, p < 0.001) showed BA had significant positive effect on SLA whereas % pine had significant negative effect on SLA in both models (Z value = -5.04, p < 0.001 in Model 1; Z value = -4.96, p < 0.001 in Model 2) (Table 5). Both variables C-N ratio and SM did not show significant effects on SLA for any of the models. Furthermore, the interaction between BA and % pine was also not significant in Model 1 (Z value = -0.88, p = 0.379) (Table 5).

Table 5. Summary of mode	l output for co	wberry old growth	in the large-scale study
--------------------------	-----------------	-------------------	--------------------------

AIC = 435.3				
Variables	Estimate	Standard Error	Z value	<b>Pr(&gt; z )</b>
Intercept	3.19	0.17	18.46	< 0.001
BA	0.65	0.16	4.07	< 0.001
% pine	-0.84	0.17	-5.04	< 0.001
C-N ratio	-0.23	0.21	-1.09	0.278
SM	-0.07	0.15	-0.45	0.655
$BA \times \%$ pine	-0.15	0.17	-0.88	0.379

*Model 1:*  $SLA = BA \times \%$  pine + C-N ratio + SM + Stand (as random intercept effect) AIC - 435.3

*Model 2: SLA* = *BA* + % *pine* + *C*-*N ratio* + *SM* + *Stand* (*as random intercept effect*) AIC = 434.1

Variables	Estimate	Standard Error	Z value	<b>Pr</b> (> z )
Intercept	3.2	0.18	18.12	< 0.001
BA	0.69	0.16	4.39	< 0.001
% pine	-0.81	0.16	-4.96	< 0.001
C-N ratio	-0.257	0.21	-1.23	0.217
SM	-0.1	0.15	-0.66	0.506

#### 3.1.1 Model prediction for large-scale study

The model prediction is demonstrated in Figure 4 for low, mean, and high % pine along with respective regression equations and R-square values. For all species, the predicted SLA increases with an increase in BA for all categories of % pine, yet the lowest predicted SLA values occur in the presence of high % pine (Figure 4). Predicted SLA ranged from 4.7-16.3 mm<sup>2</sup>/mg (under low % pine), 3.5-15.1 mm<sup>2</sup>/mg (under mean % pine), and 2.5-14.1 mm<sup>2</sup>/mg (under high % pine) (Figure 4a). For every increase of 1  $m^2$ /ha in BA, the predicted SLA of bilberry increases by 0.137 mm<sup>2</sup>/mg in all three categories of % pine (Figure 4a). In the case of cowberry new growth, the effect of interaction between BA and % pine was significant (Table 4), and hence, variations can be seen in SLA among BA gradient between three categories of % pine, with the highest for mean % pine (SLA increases by 0.15 mm<sup>2</sup>/mg for every increase of 1 m<sup>2</sup>/ha in BA) (Figure 4b). Predicted SLA ranged from 1.4-12.7 mm<sup>2</sup>/mg (under low % pine), 0.8-8.03 mm<sup>2</sup>/mg (under mean % pine), and 0.4-4.9 mm<sup>2</sup>/mg (under high % pine) for cowberry new growth (Figure 4b). For cowberry old growth, predicted SLA ranged from 2.7-6.9 mm<sup>2</sup>/mg (under low % pine), 3.1-5.8 mm<sup>2</sup>/mg (under mean pine), and 0.8-5 mm<sup>2</sup>/mg (under high % pine) (Figure 4c). Also, the R-square value is high for each case indicating the strong association between predicted SLA values and BA.



Figure 4. Predicted SLA for BA in the large-scale study using best model for a) Bilberry b) Cowberry new growth, c) Cowberry old growth. Model 1 (i.e., with the interaction between BA and % pine) resulted in the best fit for cowberry new growth. Model 2 (i.e., without interaction between BA and % pine) resulted in the best fit for bilberry and cowberry old growth. Grey regions represent a 95% confidence interval. Red, blue, and black colour indicates high, mean, and low values of % pine, respectively. Note the difference in y-axis values between the three figures.

### 3.2 Mixed forest study

For bilberry, Model 2 (AIC = 236.8) was slightly better than Model 1 (AIC = 238.2). Both models showed that BA has a significant positive effect on SLA (Z value = 4.68, p < 0.001 in Model 1; Z value = 4.58, p < 0.001 in Model 2) (Table 6). On the other hand, % broadleaves had a significant negative effect on SLA (Z value = -2.77, p < 0.001 in Model 1; Z value = -2.78, p < 0.001 in Model 2) (Table 6). Likewise, SM also had a significant negative effect on SLA in both models (Z value = -2.61, p < 0.001 in Model 1; Z value = -2.54, p < 0.05 in Model 2) (Table 6). However, the C-N ratio did not show significant effects on SLA for both models. Furthermore, the interaction between BA and % broadleaves was not significant (Z value = -0.81, p = 0.42) in Model 1.

Table 6. Summary of model output for bilberry in mixed forest study.

AIC = 238.2				
Variables	Estimate	Standard	Z value	<b>Pr</b> (>  <b>z</b>  )
		Error		
Intercept	12.57	0.63	19.87	< 0.001
BA	3.5	0.75	4.68	< 0.001
% broadleaves	-2.27	0.82	-2.77	< 0.001
C-N ratio	-0.18	0.7	-0.25	0.801
SM	-1.9	0.73	-2.61	< 0.001
$BA \times \%$ broadleaves	-0.72	0.9	-0.81	0.42

*Model 1:*  $SLA = BA \times \%$  broadleaves + C-N ratio + SM

*Model 2: SLA* = *BA* + % *broadleaves* + *C*-*N ratio* + *SM* 

AIC = 236.8				
Variables	Estimate	Standard	Z value	<b>Pr(&gt; z )</b>
		Error		
Intercept	12.61	0.64	19.82	< 0.001
BA	3.37	0.74	4.58	< 0.001
% broadleaves	-1.9	0.68	-2.78	< 0.001
C-N ratio	-0.11	0.7	-0.16	0.874
SM	-1.86	0.73	-2.54	0.011

In cowberry new growth, Model 2 (AIC = 494.4) was better than Model 1 (AIC = 496.4). BA (Z value = 3.03, p < 0.001 in Model 1; Z value = 3.05, p < 0.001 in Model 2) had a significant positive effect on SLA whereas % broadleaves (Z value = -2.31, p < 0.05 in Model 1; Z value = -2.65, p < 0.001 in Model 2) had a significant negative effect on SLA (Table 7). The C-N ratio and SM did not show significant effects on SLA for any models. Furthermore, the interaction between BA and % broadleaves was not significant in Model 1 (Z value = -0.21, p = 0.835) (Table 7).

Table 7. Summary of model output for cowberry new growth in mixed forest study.

AIC = 496.4				
Variables	Estimate	Standard	Z value	<b>Pr(&gt; z )</b>
		Error		
Intercept	9.36	0.55	16.91	< 0.001
BA	1.98	0.65	3.03	< 0.001
% broadleaves	-1.66	0.72	-2.31	0.021
C-N ratio	0.04	0.61	0.07	0.948
SM	-1.24	0.63	-1.95	0.051
$BA \times \%$ broadleaves	-0.16	0.78	-0.21	0.835

*Model 1:*  $SLA = BA \times \%$  broadleaves + C-N ratio + SM

Model 2: SLA = BA + % broadleaves + C-N ratio + SM

AIC = 494.4				
Variables	Estimate	Standard	Z value	<b>Pr(&gt; z )</b>
		Error		
Intercept	9.36	0.55	16.96	< 0.001
BA	1.95	0.64	3.05	< 0.001
% broadleaves	-1.57	0.59	-2.65	< 0.001
C-N ratio	0.05	0.61	0.09	0.928
SM	-1.23	0.64	-1.94	0.052

Model 2 (AIC = 125) was slightly better than Model 1 (AIC = 126.9) for cowberry old growth. BA had a significant positive effect on SLA (Z value = 2.67, p < 0.001 in Model 1; Z value = 2.79, p < 0.001 in Model 2), indicating that an increase in BA was associated with higher SLA values (Table 8). On the other hand, % broadleaves did not have a significant effect on SLA in Model 1, whereas in Model 2 it had a significant negative effect (Z value = -2.06, p < 0.05), suggesting that an increase in the proportion of broadleaves was associated with lower SLA values (Table 8). Yet, C-N ratio, SM, and interaction between BA and % broadleaves did not have significant effects on SLA in Model 1. Likewise, SLA was not affected by the C-N ratio or SM in Model 2.

Table 8. Summary of model output for cowberry old growth in mixed forest study.

AIC = 126.9				
Variables	Estimate	Standard	Z value	<b>Pr(&gt; z )</b>
		Error		
Intercept	2.81	0.16	17.86	< 0.001
BA	0.5	0.19	2.67	< 0.001
% broadleaves	-0.32	0.2	-1.57	0.116
C-N ratio	-0.15	0.17	-0.87	0.382
SM	-0.03	0.18	-0.21	0.832
$BA \times \%$ broadleaves	0.05	0.22	0.24	0.809

**Model 1**:  $SLA = BA \times \%$  broadleaves + C-N ratio + SM

Model 2: SLA = BA + % broadleaves + C-N ratio + SM

AIC = 125				
Variables	Estimate	Standard	Z value	<b>Pr</b> (>  <b>z</b>  )
		Error		
Intercept	2.81	0.16	17.88	< 0.001
BA	0.51	0.18	2.79	< 0.001
% broadleaves	-0.35	0.17	-2.06	0.039
C-N ratio	-0.16	0.17	-0.91	0.364
SM	-0.04	0.18	-0.23	0.819

#### 3.2.1 Model prediction for mixed forest study

The predicted values of SLA for bilberry ranged from 7.9-23.5 mm<sup>2</sup>/mg (under low % broadleaves), 5.2-20.9 mm<sup>2</sup>/mg (under mean % broadleaves), and 1.5-17.15 mm<sup>2</sup>/mg (under high % broadleaves) (Figure 5a). Further, predicted values of SLA for cowberry new growth ranged from 6.9-16.5 mm<sup>2</sup>/mg (under low % broadleaves), 4.7-14.3 mm<sup>2</sup>/mg (under mean % broadleaves), and 1.7-11.3 mm<sup>2</sup>/mg (under high % broadleaves) (Figure 5b). Furthermore, SLA for cowberry old growth ranged from 2.1-4.2 mm<sup>2</sup>/mg (under low % broadleaves), 1.7-3.8 mm<sup>2</sup>/mg (under mean % broadleaves), and 0.9-3.1 mm<sup>2</sup>/mg (under high % broadleaves) (Figure 5c). For every 1 m<sup>2</sup>/ha increase in BA, the increase in the SLA of bilberry was by 0.5 m<sup>2</sup>/mg in all categories of % broadleaves (as shown by the regression equation in Figure 5a). For cowberry new growth the increase in SLA for every 1 m<sup>2</sup>/ha increase in BA was 0.3 mm<sup>2</sup>/mg (Figure 5b) whereas, for cowberry old growth, this value was only 0.07 m<sup>2</sup>/ha (Figure 5c).



Figure 5. Predicted SLA for BA in the mixed forest study using best model for a) Bilberry b) Cowberry new growth, c) Cowberry old growth. Model 2 (i.e., without interaction between BA and % broadleaves) resulted in the best fit for all three. Grey regions represent a 95% confidence interval. Red, blue, and black colour indicates high, mean, and low values of % broadleaves, respectively. Note the difference in y-axis values between the three figures.

### 3.3 Transplant experiment

The best model for wild strawberry was Model 2 (Table 9). BA was highly significant in all models with positive estimates indicating a strong positive relation between BA and SLA (Table 9). The percentages of broadleaves were significant in Model 1 (Z value = -3.65, p < 0.001) and Model 2 (Z value = -3.56, p < 0.001) indicating the negative association between % broadleaves and SLA (Table 9). Besides, SLA did not vary significantly across different climatic origins (Model 3 and Model 4 in Table 9). Also, the effect of the interaction between BA and % broadleaves was insignificant, which indicates the relationship between SLA and BA is not modified by % broadleaves (Model 1 in Table 9). Likewise, the effect of interaction between BA and origin was also insignificant (Model 3 in Table 9), which indicates the relationship between SLA and BA is the same regardless of cold or local origin.

Table 9. Summary of model output for wild strawberry.

*Model 1*:  $SLA = BA \times \%$  broadleaves + Site (as a random intercept effect)

AIC = 694.3				
Variables	Estimate	Standard	Z	<b>Pr</b> (>  <b>z</b>  )
		Error	value	
Intercept	24.27	0.90	26.83	< 0.001
BA	2.56	0.43	5.88	< 0.001
% broadleaves	-1.49	0.41	-3.65	< 0.001
$BA \times \%$ broadleaves	-0.35	0.47	-0.75	0.453

*Model 2*:*SLA* = *BA* + % *broadleaves* + *Site* (*as a random intercept effect*)

AIC = 692.8

Variables	Estimate	Standard	Ζ	<b>Pr(&gt; z )</b>
		Error	value	
Intercept	24.32	0.87	27.96	< 0.001
BA	2.66	0.41	6.44	< 0.001
% broadleaves	-1.43	0.40	-3.56	< 0.001

*Model 3*:  $SLA = BA \times Origin + Site$  (as a random intercept effect)

AIC = 706.4				
Variables	Estimate	Standard	Ζ	<b>Pr(&gt; z )</b>
		Error	value	
Intercept	24.51	1.05	23.26	< 0.001
BA	2.65	0.6	4.45	< 0.001
Origin (warmer location)	-0.39	0.83	-0.47	0.639
BA × Origin (warmer location)	0.46	0.83	0.56	0.576

*Model 4*: *SLA* = *BA* + *Origin* + *Site* (*as a random intercept effect*)

AIC = 704.7				
Variables	Estimate	Standard	Z	<b>Pr</b> (>  <b>z</b>  )
		Error	value	
Intercept	24.51	1.05	23.25	< 0.001
BA	2.88	0.43	6.73	< 0.001
Origin (warmer location)	-0.39	0.83	-0.47	0.641

For goldenrod, the AIC value was 327.6, 326.2, 335.6, and 333.8 for Model 1, Model 2, Model 3, and Model 4 respectively, indicating Model 2 was much better than Model 3 and Model 4, and slightly better than Model 1. BA was highly significant in all models (Z value = 5.14, p < 0.001 in Model 1; Z value = 5.17, p < 0.001 in Model 2; Z value = 4.05, p < 0.001 in Model 3; Z value = 5.37, p < 0.001 in Model 4), and with positive estimates, indicating strong positive relation between BA and SLA (Table 10). On the other hand, % broadleaves had a significant negative effect in Model 1 (Z value = -2.57, p = 0.01) and Model 2 (Z

value = -2.84, p = 0.005) (Table 10). The effect of origin on SLA was not significant. Further, the effect of the interaction between BA and origin was not significant, which reflects that the effect of BA on SLA is not affected by warm or local origin (Model 3 in Table 10). Likewise, the effect of BA on SLA is not affected by % broadleaves (Model 1 in Table 10).

Table 10. Summary of model output for goldenrod.

<b>Model 1</b> : $SLA = BA \times \%$ broadleaves + Site (as a random intercept effect)					
<i>AIC</i> = 327.6					
Variables	Estimate	Standard	Z value	<b>Pr</b> (>  <b>z</b>  )	
		Error			
Intercept	21.55	0.82	26.28	< 0.001	
BA	2.56	0.5	5.14	< 0.001	
% broadleaves	-1.22	0.47	-2.57	0.010	
$BA \times \%$ broadleaves	0.42	0.53	0.79	0.428	

*Model 2: SLA* = *BA* + % *broadleaves* + *Site* (*as a random intercept effect*) AIC = 326.2

Variables	Estimate	Standard	Z value	<b>Pr(&gt; z )</b>
		Error		
Intercept	21.48	0.85	25.29	< 0.001
BA	2.42	0.47	5.17	< 0.001
% broadleaves	-1.31	0.46	-2.84	0.005

*Model 3*:  $SLA = BA \times Origin + Site$  (as a random intercept effect)

AIC = 335.6				
Variables	Estimate	Standard	Z value	<b>Pr(&gt; z )</b>
		Error		
Intercept	21.54	1.06	20.29	< 0.001
BA	2.93	0.72	4.05	< 0.001
Origin (warmer	-0.06	0.94	-0.07	0.947
location)				
$BA \times Origin$ (warmer	-0.5	0.95	-0.53	0.597
location)				

*Model 4*: *SLA* = *BA* + *Origin* + *Site* (*as a random intercept effect*)

AIC = 333.8				
Variables	Estimate	Standard	Z value	<b>Pr</b> (>  <b>z</b>  )
		Error		
Intercept	21.53	1.06	20.33	< 0.001
BA	2.65	0.49	5.37	< 0.001
Origin (Warmer	-0.06	0.95	-0.06	0.953
location)				

#### 3.3.1 Model prediction for transplant experiment

Predicted values of SLA for wild strawberry ranged from 20.7-32.6 mm<sup>2</sup>/mg (under low % broadleaves), 18.8-30.8 mm<sup>2</sup>/mg (under mean % broadleaves), and 16.9-28.8 mm<sup>2</sup>/mg (under high % broadleaves) (Figure 6a). For goldenrod, predicted SLA values ranged from 18.2-29.1 mm<sup>2</sup>/mg (under low % broadleaves), 16.5-27.4 mm<sup>2</sup>/mg (under mean % broadleaves), and 14.7-25.6 mm<sup>2</sup>/mg (under high % broadleaves) (Figure 6b). The results show that every 1 m<sup>2</sup> increase in BA led to an increase of 0.358 mm<sup>2</sup>/mg in predicted SLA for wild strawberry (Figure 6a), and an increase of 0.325 mm<sup>2</sup>/mg in predicted SLA for goldenrod (Figure 6b).



Figure 6. Predicted SLA for BA in the transplant experiment using the best model for a) Wild strawberry and b) Goldenrod. Model 2 (i.e., without interaction between BA and % broadleaves) resulted in the best fit for both species. Grey regions represent a 95% confidence interval. Red, blue, and black colour indicates high, mean, and low values of % broadleaves, respectively. Note the difference in y-axis values between the two figures.

# 4. Discussion

The results from this study show that the predicted SLA values were within the range of previous studies as documented in an open database showing traits of the Northwest European flora (LEDA- trait database) (Kleyer et al. 2008). This alignment between the predicted SLA values and SLA values from the LEDA database justifies that the prediction of the models in this study is reliable. SLA is often used as an indicator of how costly is it for plants to produce leaves (Villar & Merino 2001; Villar et al. 2021). For instance, a study by Villar and Merino (2001) in fourteen contrasting environmental conditions (from desert to rainforest), and with different life forms, shows that species with higher SLA tend to allocate less resources to leaf construction, resulting in lower leaf production costs. The models developed hence can thus be used in intraspecific SLA prediction and to analyse the production cost of leaves for the studied species, as previous studies for intraspecific SLA prediction models are rare. Nevertheless, it is essential to consider that intraspecific SLA variation (variation in SLA among individuals of the same species) can be influenced by habitat heterogeneity resulting from abiotic factors such as light, temperature, SM, and nutrient availability (Poorter et al. 2009; Kemppinen & Niittynen 2022). This study shows that the mixed forest study's SLA range for bilberry and cowberry new growth is noticeably higher than in the large-scale study (see model predictions in the results). The difference in the SLA value range can be attributed to variations in the time of leaf sample collection as the leaf samples from the mixed forest study were collected early in the summer season. Early summer is a time of active growth of leaves and as a result, plants devote more resources to enhancing photosynthesis which is brought by an increase in SLA (Chelli et al. 2019; Anderson et al. 2020). Concurrently, trees also develop their canopy during this time and increase shade. The synergistic effect created by the active growth phase and shade due to overstorey may result in high SLA in the mixed forest study. Contrarily, the mixed forest study's SLA range for cowberry old growth is lower compared to the large-scale study. This could be because plants do not allocate enough resources to old leaves for photosynthesis and leaf growth, and the response of SLA maybe only due to the shade of overstorey which was higher in large-scale study. Additionally, the SLA values of the cowberry new growth were noticeably higher than those of the cowberry old growth (see

results). This difference can be linked to the tendency of newly formed leaves to allocate more resources to photosynthesis and growth. For example, Karlsson (1985) found that the photosynthetic capacity of cowberry leaves in the second year was two-thirds of the value observed in the first year. This may be due to the beginning of leaf senescence in older leaves. Further older leaves may have started producing more carbon-rich compounds which results in thicker leaf tissues and hence low SLA as noted by some researchers in tree species (Reich et al. 1991; Milla et al. 2008). Moreover, biotic interactions, such as competition, facilitation, or amensalism, may affect species differently along the climatic gradient (Balandier et al. 2022b), which can contribute to trait variations.

### 4.1 Effect of overstorey structure and composition

Overstorey vegetation structure can either compete against or facilitate the understorey vegetation growth, influencing its SLA by controlling light transmittance and soil organic matter availability through leaf litterfall. The findings of this study demonstrate that forest density (as represented by BA) and overstorey composition (as represented by % pine or % broadleaves) are important predictors of SLA in understorey plant species. These findings are consistent with previous research (Chelli et al. 2019; Vanneste et al. 2019) and confirm the role of overstorey structure and composition in influencing the SLA of understorey plant species. Specifically, a denser overstorey limits the amount of light entering the forest understorey (Greiser et al. 2018; Govaert et al. 2020; De Pauw et al. 2022), and thereby influencing the forest microclimate (Thom et al. 2020; Deng et al. 2023). Under such conditions of low light availability, understorey plants respond by increasing their SLA (Pérez-Harguindeguy et al. 2016). The function of light transmittance can also vary based on overstorey species composition. With an increase in the percentage of pine, the SLA of the understorey decreases (see large-scale study results) and with an increase in the percentage of broadleaves, the SLA of the understorey decreases (see results from the mixed forest study and the transplant experiment). This observed decrease in SLA with an increasing percentage of pine (large-scale study results) or broadleaves (results from the mixed forest study and the transplant experiment) can be attributed to the fact that pine/ broadleaves transmit more light in the understorey as compared to Norway spruce (Petersson et al. 2019). Further broadleaves shed their leaves during winter allowing more entry of light and a wider time window in early spring for understorey plants to accumulate resources before leaf-out and canopy closure.

Likewise, a higher density of overstorey trees can lead to increased nutrient availability in the soil through increased litterfall and nutrient cycling. In fact, activities that reduce forest density are found to reduce soil organic contents (Barbier et al. 2008; Yang et al. 2022) whereas a positive correlation between litter biomass and forest density has been observed (Pérez-alavez et al. 2023). Moreover, the composition of the overstorey can influence litter biomass. Broadleaves, being deciduous and shedding leaves in winter, contribute more litterfall and facilitate the accumulation of soil organic compounds compared to pine (Barbier et al. 2008; Sheffer et al. 2015; Pérez-alavez et al. 2023). Consequently, it increases soil fertility in broadleaved forests and enhances the conditions for understorey growth, resulting in higher SLA values compared to coniferous forests. Yet, below ground competition for soil resources between overstorey and understorey has been documented in past studies (Barbier et al. 2008; Balandier et al. 2022b) which will ultimately decrease SLA. Interestingly, even in the transplant experiment where the soil was not influenced by litterfall, the SLA of understorey plants increased with forest density, indicating that light has a stronger effect than soil organic matter, as supported by previous studies (Poorter et al. 2009, 2019; De Pauw et al. 2022).

In addition, the results indicate that there is no interaction between the variable BA and the % pine (in the large-scale study), and between the variable BA and the % broadleaves (in the mixed forest study and the transplant experiment). This indicates that the effect of forest density on SLA generally remains consistent regardless of overstorey species composition. However, a notable exception is observed in the case of cowberry new growth in the large-scale study, indicating that the impact of forest density on the SLA of cowberry's new growth is influenced by the presence of pine. Cowberry, particularly its new leaves during the early growth phase, may require increased light exposure due to their light demanding nature (Ritchie 1955) and active growth phase, and hence might have responded differently. Consequently, different combinations of forest density and pine presence can cause variations in light transmittance in the forest understorey, leading to significant variations in the SLA of cowberry new growth. For instance, in dense forests where the proportion of pine is high, the light conditions can still be better than in similar dense forests with Norway spruce (Petersson et al. 2019), leading to lower SLA values of cowberry new growth than under Norway spruce forest. Nevertheless, stand structure has been often found to strongly influence light transmittance more than tree composition (Kovács et al. 2017). Therefore, forest structure is a major variable affecting SLA on understorey plants.

### 4.2 Effect of soil moisture and carbon-nitrogen ratio

A significant effect of SM and C-N ratio on the SLA of understorey plants was not observed except for two cases: in the large-scale study, where the SLA of cowberry new growth was negatively affected by SM (Model 2) and in the mixed forest study, where the SLA of bilberry was negatively affected by SM (Model 1 and Model 2). However, a study in western Canadian forest understorey plant communities suggested that SM, and soil C-N ratio were key factors influencing interspecific SLA variation (Shovon et al. 2019). The study revealed that higher levels of SM and lower C-N ratio were associated with a shift towards more acquisitive resource-use strategies, which was reflected in an increase in SLA. Yet, this study was focused on intraspecific SLA variation which might have resulted in contrasting results. Nevertheless, SM was found to be an important predictor of SLA in an arctic region with a positive relationship for bilberry and goldenrod, while no relationship was found for cowberry (Kemppinen & Niittynen 2022). However, the difference in the direction of intraspecific trait response along climatic gradients has been documented in previous research (Roybal & Butterfield 2019). Moreover, a piece of research conducted by compiling datasets on SLA from various systems and growth forms demonstrated a negative relationship between SLA and soil C-N ratio, reflecting that decreasing soil C-N ratio corresponded to increased nutrient availability and higher SLA (Ordoñez et al. 2009) yet the results from this study did not find any significant relationship with soil C-N ratio. The same study suggested that up to 32-34% of the variation in SLA among sites could be explained solely by the soil C-N ratio. Differences in plant species and environmental conditions examined could be another reason accounting for the disparities between the result from this study and previous studies (Gong & Gao 2019). Further, a study with alder and spruce trees in boreal forests shows that SM has a negative linear relationship with SLA and that higher soil C-N values are correlated with lower SLA values (Anderson et al. 2020). The same research indicated that the relationship between SM and SLA can be non-linear showing the possibility that there is an optimum SM level where SLA is highest and then decreases with further increases in SM (Anderson et al. 2020), which could explain the contrasting conclusions across different studies. Besides, variations in measurement methods or analytical techniques could have influenced the results.

The complex interactions between environmental factors and plant traits make it difficult to draw conclusive findings on this relationship. Further research is needed to better understand the relationship between SM, C-N ratio, and understorey SLA at a global scale, as this has important implications for predicting understorey responses to how understorey responds to the changing environment. In addition, climatic variables (temperature and precipitation) can affect SM by affecting the evaporation and evapotranspiration processes, as well as driving the C-N ratio by affecting microbial biomass (Woo & Seo 2022). Hence, understanding how these variables in combination with forest overstorey affect the SLA of different understorey species under different environmental conditions is essential considering the changes that will be brought about by climate change.

### 4.3 Effect of warming

The results from the transplant experiment show that the SLA of understorey wild strawberry and goldenrod did not differ significantly based on their origin (i.e., cold and local origin). This result can be rationalized by two possible reasons. Firstly, there is no evidence of divergent SLA among understorey plants, as no pre-transplant material from the cold origin was available for comparison. While specific research for the same species is lacking, previous studies have demonstrated a general increase in SLA with an increase in latitudinal gradient (De Frenne et al. 2013; Sun et al. 2023), weakening the argument that SLA of wild strawberry and goldenrod originating from different regions were same. Secondly, the SLA of wild strawberry and goldenrod of cold origin may have shown rapid adaptation and reached a comparable level of SLA with the understorey plants of local origin assuming the SLA of plants of two origins varies significantly before transplantation. Such a rapid response of understorey plants may have been facilitated by the plastic nature of SLA (Liu et al. 2016). Consequently, this suggests that the SLA of the understorey plants will adapt to climate change induced temperature rise. The change in SLA as a response to adaptation to climate induced temperature rise can be supported based on previous research which shows SLA of some understorey species increases (De Kort et al. 2020) or decreases (Blondeel et al. 2020; Maes et al. 2020) with warming treatments. Contrarily, other research shows SLA did not change with experimental warming and it is light which is more important than temperature for the SLA of the understorey (De Pauw et al. 2022). Yet the transplant experiment design was made such that both warm and cold regions had similar overstorey density and species composition so that the plots receive similar light amount, indicating that understorey wild strawberry and goldenrod of cold origin changed their SLA in response to temperature warming. Furthermore, the interaction between BA (which represents forest density) and origin in the transplant experiment was not significant which implies that there is not a difference in adaptation of understorey to light between cold and local origin. Hence, even if climate change will shift the overstorey density, SLA of wild strawberry and goldenrod would not be affected significantly.

Different species may exhibit different response patterns for SLA (Liu et al. 2023) and conducting the same experiment with different species could yield contrasting results. A study in 76 natural ecosystems of China found that SLA (of various life forms: herbs, shrubs, trees) was positively correlated with mean annual temperature, yet this relationship changed when the temperature was either

below or above a certain threshold (20°C in this case) (Liu et al. 2023). The study also found that SLA has a limited range of variation. To adapt to changes brought by climate, SLA would need to potentially change by 12% (Liu et al. 2023). Moreover, the same study found that SLA is not easily passed down genetically from one generation to the next (Liu et al. 2023). Another factor to consider is that species tend to react at a slower velocity than the observed velocity of climate change (a phenomenon called climatic debt) (Zellweger et al. 2020; Richard et al. 2021). Even in the transplant experiment, wild strawberry and goldenrod were moved from their cold region to a warmer region in a way that the temperature difference that they experienced was much quicker than what would be expected under natural warming conditions. However, to account for this factor in the transplant experiment, understorey plants were left for a few seasons for acclimatization to new environmental conditions.

Furthermore, it should also be noted that uniform soil conditions and protection from herbivores in the transplant experiment might have inadvertently reduced the variability in the plants' environmental conditions, which might not be the case under normal conditions in forests (Nooten & Hughes 2017). Moreover, a possible caveat is that the transplant experiment study only tested for differences in SLA between two specific climate regimes (warm and cold) and did not investigate how plants might respond to more extreme or unpredictable climate conditions.

#### 4.4 Implications for forest management

Forest management is critical in the context of global climate and land-use change and its potential effects on understorey vegetation. Controlling the density of a forest stand, we can control light, provide temperature buffering, and mitigate the ongoing effects of climate change on the understorey (Balandier et al. 2022b; Christiansen et al. 2022; Deng et al. 2023). Especially, the density of Norway spruce plantations is prominent in Sweden (Felton et al. 2022). As observed in this study the increased density of the forest will also increase the SLA of bilberry, cowberry, wild strawberry, and goldenrod. High SLA values correspond to higher litter decomposition rates (Garnier et al. 2004; Mokany et al. 2008), directly influencing nutrient cycling, a pivotal ecosystem process. An increase in the SLA of the understorey is also linked to enhanced ecosystem services of water regulation, biomass production, and aesthetic appeal (Shome et al. 2023). On the other hand, berry production and growth of other understorey organs can decrease with an increase of SLA as plants allocate more resources for the expansion of leaf area. The rise in SLA can also lead to increased plant palatability (Funk et al. 2017), yet plants may decrease their SLA as a response to herbivores due to the plastic nature of SLA. It is worth considering that bilberry and cowberry are

foundation species that have a significant effect on shaping local and global biodiversity, non-trophic facilitation, and regulating ecosystem dynamics (Ellison & Degrassi 2017; Borst et al. 2018; Ellison et al. 2019; Forzieri et al. 2020; O'Brien et al. 2020; Lortie et al. 2022). Therefore, modification in the SLA of these foundation plants also means a change in the services provided by them.

Global phenomena such as climate change further make the interaction between overstorey and understorey complicated as the severity of forest disturbances may increase in the future due to climate change (Seidl et al. 2014; Patacca et al. 2023). These more extreme conditions may result in significant overstorey driven changes in SLA for these species, beyond those considered in this study. Climate change induced forest disturbances may lead to tree mortality, which will likely promote canopy openness and decrease tree density in the forest (Kumar et al. 2021). Consequently, the natural functions of light control and temperature buffering provided by the forest will be compromised (Balandier et al. 2022b; Christiansen et al. 2022; Deng et al. 2023). Under conditions of a severe decrease in forest density of overstorey, the understorey would decrease its SLA. On the contrary, replanting trees in higher densities after clearcutting will result in an understorey with high SLA.

Lately, there is also advocacy towards overstorey species mixtures due to their potential for providing multiple ecosystem services (Felton et al. 2016; Huuskonen et al. 2021; Ara et al. 2022). The findings from this study also support the promotion of overstorey mixing with pine and broadleaves to moderate SLA. Notably, mixing the overstorey species that permit greater light penetration can potentially moderate SLA and its related effects, and optimize understorey growth and berry production. The light transmittance, temperature, and resource availability in forest understorey can be optimized by mixing overstorey species with different structures and traits (Zhang et al. 2022).

# 5. Conclusion and Recommendations

Overall, this study developed reliable SLA prediction models for bilberry, cowberry, wild strawberry, and goldenrod which can improve our ability to predict the responses of understorey plant communities to changing environmental conditions and make better forest management decisions. The results from this study highlight that forest density positively influences the SLA of bilberry, cowberry, wild strawberry, and goldenrod. The influence of forest density is mainly through the control of light transmittance, which is an important resource affecting SLA. At the same time, the proportion of overstorey species with greater light transmittance capacity decreases the SLA of bilberry, cowberry, wild strawberry, and goldenrod. The general trend from most of the models demonstrated SM and C-N ratio did not affect SLA significantly. The interaction between forest density and the percentage of broadleaves was not significant whereas the interaction between forest density and the percentage of pine had a significant effect on cowberry new growth in large-scale study. Moreover, in the transplant experiment, SLA of wild strawberry and goldenrod did not change significantly with climatic origins. The temperature variations specific to different climatic origins did not have a significant impact on how forest density affects SLA of wild strawberry and goldenrod.

The findings from this research also open a door for future research which may involve quantifying to what extent functional traits of plants can adapt to a changing environment and even climate, and how long does it take to observe noticeable changes in SLA of understorey plants. It has also been suggested to combine large-scale studies that look at long-term adaptations of the understorey with small-scale experiments that focus on short-term acclimation (Liu et al. 2023). As intraspecific trait variations can comprise a range of values within an ecosystem, further studies should be conducted to understand the variation in trait values within species and how they are influenced by biotic interactions and environmental gradients. The mixed forest study was limited to a site that can limit the generality of the findings. Hence, a similar study in a wide latitudinal area is suggested. Conducting similar research works for other understorey species would also contribute to the addition of scientific knowledge. Additionally, studying the variation of SLA of the understorey during the whole growth season is encouraged to broaden the understanding of seasonal changes in SLA for various understorey species. It is also important that future research focuses on the interactions between SLA and soil, specifically looking at how understorey vegetation and overstorey trees affect one another.

# References

- Ali, A., Yan, E.R., Chang, S.X., Cheng, J.Y. & Liu, X.Y. (2017). Community-weighted mean of leaf traits and divergence of wood traits predict aboveground biomass in secondary subtropical forests. *Science of the Total Environment*, 574 (January), 654–662. https://doi.org/10.1016/j.scitotenv.2016.09.022
- Anderson, C.G., Bond-Lamberty, B. & Stegen, J.C. (2020). Active layer depth and soil properties impact specific leaf area variation and ecosystem productivity in a boreal forest. *PLoS ONE*, 15 (12 December), 1–18. https://doi.org/10.1371/journal.pone.0232506
- Ara, M., Felton, A.M., Holmström, E., Petersson, L., Berglund, M., Johansson, U. & Nilsson, U. (2022). Pre-commercial thinning in Norway spruce-birch mixed stands can provide abundant forage for ungulates without losing volume production. *Forest Ecology and Management*, 520 (March). https://doi.org/10.1016/j.foreco.2022.120364
- Balandier, P., Gobin, R., Prévosto, B. & Korboulewsky, N. (2022a). The contribution of understorey vegetation to ecosystem evapotranspiration in boreal and temperate forests: a literature review and analysis. *European Journal of Forest Research*, 141 (6), 979–997. https://doi.org/10.1007/s10342-022-01505-0
- Balandier, P., Mårell, A., Prévosto, B. & Vincenot, L. (2022b). Tamm review: Forest understorey and overstorey interactions: So much more than just light interception by trees. *Forest Ecology and Management*, 526 (June), 1–17. https://doi.org/10.1016/j.foreco.2022.120584
- Barbier, S., Gosselin, F. & Balandier, P. (2008). Influence of tree species on understory vegetation diversity and mechanisms involved — A critical review for temperate and boreal forests. *Forest Ecology and Management*, 254 (1), 1–15. https://doi.org/10.1016/j.foreco.2007.09.038
- Blondeel, H., Perring, M.P., Lombaerde, E. De, Depauw, L., Landuyt, D. & Govaert, S. (2020). Individualistic responses of forest herb traits to environmental change. *Plant Biology*, 22 (4), 553–749. https://doi.org/10.1111/plb.13103
- Borst, A.C.W., Verberk, W.C.E.P., Angelini, C., Schotanus, J., Wolters, J.W.,
  Christianen, M.J.A., van der Zee, E.M., Derksen-Hooijberg, M. & van der Heide,
  T. (2018). Foundation species enhance food web complexity through non-trophic facilitation. *PLoS ONE*, 13 (8), 1–15.
  https://doi.org/10.1371/journal.pone.0199152
- Brooks, M.E., Kristensen, K., van Benthem, K.J., Magnusson, A., Berg, C.W., Nielsen, A., Skaug, H.J., Mächler, M. & Bolker, B.M. (2017). glmmTMB balances speed

and flexibility among packages for zero-inflated generalized linear mixed modeling. *R Journal*, 9 (2), 378–400. https://doi.org/10.32614/rj-2017-066

- Bu, W., Huang, J., Xu, H., Zang, R., Ding, Y., Li, Y., Lin, M., Wang, J. & Zhang, C. (2019). Plant functional traits are the mediators in regulating effects of abiotic site conditions on aboveground carbon stock-evidence from a 30 ha tropical forest plot. *Frontiers in Plant Science*, 9 (Jan), 1958. https://doi.org/10.3389/fpls.2018.01958
- Burns, K.C. (2004). Patterns in specific leaf area and the structure of a temperate heath community. *Diversity and Distributions*, 10 (2), 105-112. https://doi.org/10.1111/j.1366-9516.2004.00058.x
- Chelli, S., Simonetti, E., Wellstein, C., Campetella, G., Carnicelli, S., Andreetta, A., Giorgini, D., Puletti, N., Bartha, S. & Canullo, R. (2019). Effects of climate, soil, forest structure and land use on the functional composition of the understorey in Italian forests. *Journal of Vegetation Science*, 30 (6), 1110–1121. https://doi.org/10.1111/jvs.12792
- Christiansen, D.M., Iversen, L.L., Ehrlén, J. & Hylander, K. (2022). Changes in forest structure drive temperature preferences of boreal understorey plant communities. *Journal of Ecology*, 110 (3), 631–643. https://doi.org/10.1111/1365-2745.13825
- Cornelissen, J.H.C., Lavorel, S., Garnier, E., Díaz, S., Buchmann, N., Gurvich, D.E., Reich, P.B., Ter Steege, H., Morgan, H.D., Van Der Heijden, M.G.A., Pausas, J.G. & Poorter, H. (2003). A handbook of protocols for standardised and easy measurement of plant functional traits worldwide. *Australian Journal of Botany*, 51 (4), 335–380. https://doi.org/10.1071/BT02124
- Dagan, U. & Izhaki, I. (2019). Understory vegetation in planted pine forests governs bird community composition and diversity in the Eastern mediterranean region. *Forest Ecosystems*, 6 (1), 1–16. https://doi.org/10.1186/s40663-019-0186-y
- De Frenne, P., Graae, B.J., Rodríguez-Sánchez, F., Kolb, A., Chabrerie, O., Decocq, G., De Kort, H., De Schrijver, A., Diekmann, M., Eriksson, O., Gruwez, R., Hermy, M., Lenoir, J., Plue, J., Coomes, D.A. & Verheyen, K. (2013). Latitudinal gradients as natural laboratories to infer species' responses to temperature. *Journal of Ecology*, 101 (3), 784–795. https://doi.org/10.1111/1365-2745.12074
- De Kort, H., Panis, B., Helsen, K., Douzet, R., Janssens, S.B. & Honnay, O. (2020). Preadaptation to climate change through topography-driven phenotypic plasticity. *Journal of Ecology*, 108 (4), 1465–1474. https://doi.org/10.1111/1365-2745.13365
- Deng, J., Fang, S., Fang, X., Jin, Y., Kuang, Y., Lin, F., Liu, J., Ma, J., Nie, Y., Ouyang, S., Ren, J., Tie, L., Tang, S., Tan, X., Wang, X., Fan, Z., Wang, Q.-W., Wang, H. & Liu, C. (2023). Forest understory vegetation study: current status and future trends. *Forestry Research*, 3 (1). https://doi.org/10.48130/fr-2023-0006
- De Pauw, K., Sanczuk, P., Meeussen, C., Depauw, L., De Lombaerde, E., Govaert, S., Vanneste, T., Brunet, J., Cousins, S.A., Gasperini, C. & Hedwall, P.O. (2022). Forest understorey communities respond strongly to light in interaction with forest structure, but not to microclimate warming. *New Phytologist*, 233(1), 219-235. https://doi.org/10.1111/nph.17803

- Duan, B., Xiao, R., Cai, T., Man, X., Ge, Z., Gao, M. & Mencuccini, M. (2023). Understory species composition mediates soil greenhouse gas fluxes by affecting bacterial community diversity in boreal forests. *Frontiers in Microbiology*, 13 (January). https://doi.org/10.3389/fmicb.2022.1090169
- Dunnington, D. and Thorne, B., 2022. ggspatial: Spatial Data Framework for ggplot2. *R* package version, 1(4).
- Ellison, A.M., Buckley, H.L., Case, B.S., Cardenas, D., Duque, Á.J., Lutz, J.A., Myers, J.A., Orwig, D.A. & Zimmerman, J.K. (2019). Species diversity associated with foundation species in temperate and tropical forests. *Forests*, 10 (2). https://doi.org/10.3390/f10020128
- Ellison, A.M. & Degrassi, A.L. (2017). All species are important, but some species are more important than others. *Journal of Vegetation Science*, 28 (4), 669–671. https://doi.org/10.1111/jvs.12566
- Felton, A., Felton, A.M., Wam, H.K., Witzell, J., Wallgren, M., Löf, M., Sonesson, J., Lindbladh, M., Björkman, C., Blennow, K., Cleary, M., Jonsell, M., Klapwijk, M.J., Niklasson, M., Petersson, L., Rönnberg, J., Sang, Å.O., Wrethling, F. & Hedwall, P.O. (2022). Forest biodiversity and ecosystem services from sprucebirch mixtures: The potential importance of tree spatial arrangement. *Environmental Challenges*, 6 (January 2022), 100407. https://doi.org/10.1016/j.envc.2021.100407
- Felton, A., Nilsson, U., Sonesson, J., Felton, A.M., Roberge, J.M., Ranius, T., Ahlström, M., Bergh, J., Björkman, C., Boberg, J., Drössler, L., Fahlvik, N., Gong, P., Holmström, E., Keskitalo, E.C.H., Klapwijk, M.J., Laudon, H., Lundmark, T., Niklasson, M., Nordin, A., Pettersson, M., Stenlid, J., Sténs, A. & Wallertz, K. (2016). Replacing monocultures with mixed-species stands: Ecosystem service implications of two production forest alternatives in Sweden. *Ambio*, 45, 124–139. https://doi.org/10.1007/s13280-015-0749-2
- Forzieri, G., Miralles, D.G., Ciais, P., Alkama, R., Ryu, Y., Duveiller, G., Zhang, K., Robertson, E., Kautz, M., Martens, B., Jiang, C., Arneth, A., Georgievski, G., Li, W., Ceccherini, G., Anthoni, P., Lawrence, P., Wiltshire, A., Pongratz, J., Piao, S., Sitch, S., Goll, D.S., Arora, V.K., Lienert, S., Lombardozzi, D., Kato, E., Nabel, J.E.M.S., Tian, H., Friedlingstein, P. & Cescatti, A. (2020). Increased control of vegetation on global terrestrial energy fluxes. *Nature Climate Change*, 10 (4), 356–362. https://doi.org/10.1038/s41558-020-0717-0
- Funk, J.L., Larson, J.E., Ames, G.M., Butterfield, B.J., Cavender-Bares, J., Firn, J., Laughlin, D.C., Sutton-Grier, A.E., Williams, L. & Wright, J. (2017). Revisiting the Holy Grail: Using plant functional traits to understand ecological processes. *Biological Reviews*, 92 (2), 1156–1173. https://doi.org/10.1111/brv.12275
- Garnier, E., Cortez, J., Billès, G., Navas, M.L., Roumet, C., Debussche, M., Laurent, G., Blanchard, A., Aubry, D., Bellmann, A., Neill, C. & Toussaint, J.P. (2004). Plant functional markers capture ecosystem properties during secondary succession. *Ecology*, 85 (9), 2630–2637. https://doi.org/10.1890/03-0799

- Gong, H. & Gao, J. (2019). Soil and climatic drivers of plant SLA (specific leaf area). Global Ecology and Conservation, 20, e00696. https://doi.org/10.1016/j.gecco.2019.e00696
- Govaert, S., Meeussen, C., Vanneste, T., Bollmann, K., Brunet, J., Cousins, S.A.O., Diekmann, M., Graae, B.J., Hedwall, P.O., Heinken, T., Iacopetti, G., Lenoir, J., Lindmo, S., Orczewska, A., Perring, M.P., Ponette, Q., Plue, J., Selvi, F., Spicher, F., Tolosano, M., Vermeir, P., Zellweger, F., Verheyen, K., Vangansbeke, P. & De Frenne, P. (2020). Edge influence on understorey plant communities depends on forest management. *Journal of Vegetation Science*, 31 (2), 281–292. https://doi.org/10.1111/jvs.12844
- Greiser, C., Meineri, E., Luoto, M., Ehrlén, J. & Hylander, K. (2018). Monthly microclimate models in a managed boreal forest landscape. *Agricultural and Forest Meteorology*, 250–251 (May 2017), 147–158. https://doi.org/10.1016/j.agrformet.2017.12.252
- Heilmeier, H. (2019). Functional traits explaining plant responses to past and future climate. *Flora*, 254, 1–11. https://doi.org/10.1016/j.flora.2019.04.004
- Huuskonen, S., Domisch, T., Finér, L., Hantula, J., Hynynen, J., Matala, J., Miina, J., Neuvonen, S., Nevalainen, S., Niemistö, P., Nikula, A., Piri, T., Siitonen, J., Smolander, A., Tonteri, T., Uotila, K. & Viiri, H. (2021). What is the potential for replacing monocultures with mixed-species stands to enhance ecosystem services in boreal forests in Fennoscandia? *Forest Ecology and Management*, 479 (June 2020), 118558. https://doi.org/10.1016/j.foreco.2020.118558
- Karlsson, P.S. (1985). Photosynthetic characteristics and leaf carbon economy of a deciduous and an evergreen dwarf shrub: Vaccinium uliginosum L. and V. vitis-idaea L. *Ecography*, 8 (1), 9–17. https://doi.org/10.1111/j.1600-0587.1985.tb01147.x
- Kemppinen, J. & Niittynen, P. (2022). Microclimate relationships of intraspecific trait variation in sub-Arctic plants. *Oikos*, 2022 (12), e09507. https://doi.org/10.1111/oik.09507
- Kleyer, M., Bekker, R.M., Knevel, I.C., Bakker, J.P., Thompson, K., Sonnenschein, M., Poschlod, P., Van Groenendael, J.M., Klimeš, L., Klimešová, J., Klotz, S., Rusch, G.M., Hermy, M., Adriaens, D., Boedeltje, G., Bossuyt, B., Dannemann, A., Endels, P., Götzenberger, L., Hodgson, J.G., Jackel, A.K., Kühn, I., Kunzmann, D., Ozinga, W.A., Römermann, C., Stadler, M., Schlegelmilch, J., Steendam, H.J., Tackenberg, O., Wilmann, B., Cornelissen, J.H.C., Eriksson, O., Garnier, E. & Peco, B. (2008). The LEDA Traitbase: A database of life-history traits of the Northwest European flora. *Journal of Ecology*, 96 (6), 1266–1274. https://doi.org/10.1111/j.1365-2745.2008.01430.x
- Katabuchi, M. (2015). LeafArea: an R package for rapid digital image analysis of leaf area. *Ecological Research*, *30*, 1073-1077.
- Kovács, B., Tinya, F. & Ódor, P. (2017). Stand structural drivers of microclimate in mature temperate mixed forests. *Agricultural and Forest Meteorology*, 234–235, 11–21. https://doi.org/10.1016/j.agrformet.2016.11.268

- Kumar, T., Islam, A. & Uddin, J. (2021). Forest carbon stocks under three canopy densities in Sitapahar natural forest reserve in Chittagong Hill Tracts of Bangladesh. *Forest Ecology and Management*, 492, 119217. https://doi.org/10.1016/j.foreco.2021.119217
- Kuuluvainen, T., Angelstam, P., Frelich, L., Jõgiste, K., Koivula, M., Kubota, Y., Lafleur, B. & Macdonald, E. (2021). Natural disturbance-based forest management: Moving beyond retention and Continuous-Cover Forestry. *Frontiers in Forests and Global Change*, 4 (April), 1–16. https://doi.org/10.3389/ffgc.2021.629020
- Liu, G., Wang, L., Jiang, L., Pan, X., Huang, Z., Dong, M. & Cornelissen, J.H.C. (2018). Specific leaf area predicts dryland litter decomposition via two mechanisms. *Journal of Ecology*, 106 (1), 218–229. https://doi.org/10.1111/1365-2745.12868
- Liu, Y., Dawson, W., Prati, D., Haeuser, E., Feng, Y. & Van Kleunen, M. (2016). Does greater specific leaf area plasticity help plants to maintain a high performance when shaded? *Annals of Botany*, 118 (7), 1329–1336. https://doi.org/10.1093/aob/mcw180
- Liu, Z., Zhao, M., Zhang, H., Ren, T., Liu, C. & He, N. (2023). Divergent response and adaptation of specific leaf area to environmental change at different spatio-temporal scales jointly improve plant survival. *Global Change Biology*, 29 (4), 1144–1159. https://doi.org/10.1111/gcb.16518
- Lortie, C.J., Filazzola, A., Westphal, M. & Butterfield, H.S. (2022). Foundation plant species provide resilience and microclimatic heterogeneity in drylands. *Scientific Reports*, 12 (1), 1–8. https://doi.org/10.1038/s41598-022-22579-1
- Maes, S.L., Perring, M.P., Depauw, L., Bernhardt-Römermann, M., Blondeel, H., Brūmelis, G., Brunet, J., Decocq, G., den Ouden, J., Govaert, S., Härdtle, W., Hédl, R., Heinken, T., Heinrichs, S., Hertzog, L., Jaroszewicz, B., Kirby, K., Kopecký, M., Landuyt, D., Máliš, F., Vanneste, T., Wulf, M. & Verheyen, K. (2020). Plant functional trait response to environmental drivers across European temperate forest understorey communities. *Plant Biology*, 22 (3), 410–424. https://doi.org/10.1111/plb.13082
- Mcdowell, N.G., Allen, C.D., Anderson-Teixeira, K., Aukema, B.H., Bond-Lamberty, B., Chini, L., Clark, J.S., Dietze, M., Grossiord, C., Hanbury-Brown, A., Hurtt, G.C., Jackson, R.B., Johnson, D.J., Kueppers, L., Lichstein, J.W., Ogle, K., Poulter, B., Pugh, T.A.M., Seidl, R., Turner, M.G., Uriarte, M., Walker, A.P. & Xu, C. (2020). Pervasive shifts in forest dynamics in a changing world. *Science*, 368 (6494). https://doi.org/10.1126/science.aaz9463
- Milla, R., Reich, P.B., Niinemets, Ü. & Castro-Díez, P. (2008). Environmental and developmental controls on specific leaf area are little modified by leaf allometry. *Functional Ecology*, 22 (4), 565–576. https://doi.org/10.1111/j.1365-2435.2008.01406.x
- Mokany, K., Ash, J. & Roxburgh, S. (2008). Functional identity is more important than diversity in influencing ecosystem processes in a temperate native grassland. *Journal of Ecology*, 96 (5), 884–893. https://doi.org/10.1111/j.1365-2745.2008.01395.x

- Niklas, K.J., Cobb, E.D., Niinemets, Ü., Reich, P.B., Sellin, A., Shipley, B. & Wright, I.J. (2007). "Diminishing returns" in the scaling of functional leaf traits across and within species groups. *Proceedings of the National Academy of Sciences of the United States of America*, 104 (21), 8891–8896. https://doi.org/10.1073/pnas.0701135104
- Nilsson, M. & Wardle, D.A. (2005). Understory vegetation as a forest ecosystem driver : evidence from the northern Swedish boreal forest. *Frontiers in Ecology and the Environment*, 3 (8), 421–428. https://doi.org/10.1890/1540-9295(2005)003[0421:UVAAFE]2.0.CO;2
- Nooten, S.S. & Hughes, L. (2017). The power of the transplant: direct assessment of climate change impacts. *Climatic Change*, 144 (2), 237–255. https://doi.org/10.1007/s10584-017-2037-6
- O'Brien, M.J., Carbonell, E.P., Losapio, G., Schlüter, P.M. & Schöb, C. (2020). Foundation species promote local adaptation and fine-scale distribution of herbaceous plants.pdf. *Journal of Ecology*, 109 (1), 191–203. https://doi.org/10.1111/1365-2745.13461
- Ordoñez, J.C., Van Bodegom, P.M., Witte, J.P.M., Wright, I.J., Reich, P.B. & Aerts, R. (2009). A global study of relationships between leaf traits, climate and soil measures of nutrient fertility. *Global Ecology and Biogeography*, 18 (2), 137– 149. https://doi.org/10.1111/j.1466-8238.2008.00441.x
- Patacca, M., Lindner, M., Esteban, M., Cordonnier, T., Fidej, G., Gardiner, B., Hauf, Y., Nabuurs, J., Nagel, T.A., Nikinmaa, L., Panyatov, M., Socha, J., Thom, D., Vuletic, D., Zudin, S. & Schelhaas, M.-J. (2023). Significant increase in natural disturbance impacts on European forests since 1950. *Global Change Biology*, 29 (5), 1359–1376. https://doi.org/10.1111/gcb.16531
- Pérez-Alavez, Y., Rodríguez-Ortiz, G., Santiago-García, W., Campos-Angeles, G.V., Valle, J.R.E., Martin, M.P., & Forest, L. (2023). Effect of thinning intensity on litterfall biomass and nutrient deposition in a naturally regenerated Pinus pseudostrobus Lind . forest. *Journal of Sustainable Forestry*, 42 (1), 77–94. https://doi.org/10.1080/10549811.2021.1946410
- Pérez-Harguindeguy, N., Díaz, S., Garnier, E., Lavorel, S., Poorter, H., Jaureguiberry, P., Cornwell, W.K., Craine, J.M., Gurvich, D.E., Urcelay, C., Veneklaas, E.J., Reich, P.B., Poorter, L., Wright, I.J., Ray, P., Enrico, L., Pausas, J.G., Vos, A.C. De, Buchmann, N., Funes, G., Quétier, F., Hodgson, J.G., Thompson, K., Morgan, H.D., Steege, H., Heijden, M.G.A. Van Der, Sack, L., Blonder, B., Poschlod, P., Vaieretti, M. V, Conti, G., Staver, A.C., Aquino, S. & Cornelissen, J.H.C. (2016). Leaf traits. In: *New handbook for standardised measurement of plant functional traits worldwide*. CSIRO. 186. https://doi.org/10.1071/BT12225\_CO
- Petersson, L., Holmström, E., Lindbladh, M. & Felton, A. (2019). Tree species impact on understory vegetation: Vascular plant communities of Scots pine and Norway spruce managed stands in northern Europe. *Forest Ecology and Management*, 448 (March), 330–345. https://doi.org/10.1016/j.foreco.2019.06.011

- Poorter, H., Niinemets, Ü., Ntagkas, N., Siebenkäs, A., Mäenpää, M., Matsubara, S. & Pons, T.L. (2019). A meta-analysis of plant responses to light intensity for 70 traits ranging from molecules to whole plant performance. *New Phytologist*, 223 (3), 1073–1105. https://doi.org/10.1111/nph.15754
- Poorter, H., Niinemets, Ü., Poorter, L., Wright, I.J. & Villar, R. (2009). Causes and consequences of variation in leaf mass per area (LMA): A meta-analysis. *New Phytologist*, 182 (3), 565–588. https://doi.org/10.1111/j.1469-8137.2009.02830.x
- R Core Team (2022). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL: https://www.Rproject.org/.
- Reich, P.B., Uhl, C., Walters, M.M. & Ellsworth, D.S. (1991). Leaf lifespan as a determinant of leaf structure and function among 23 Amazonian tree species. *Oecologia*, 86 (11), 16–24. https://doi.org/10.1007/BF00317383
- Richard, B., Dupouey, J.L., Corcket, E., Alard, D., Archaux, F., Aubert, M., Boulanger, V., Gillet, F., Langlois, E., Macé, S., Montpied, P., Beaufils, T., Begeot, C., Behr, P., Boissier, J.M., Camaret, S., Chevalier, R., Decocq, G., Dumas, Y., Eynard-Machet, R., Gégout, J.C., Huet, S., Malécot, V., Margerie, P., Mouly, A., Paul, T., Renaux, B., Ruffaldi, P., Spicher, F., Thirion, E., Ulrich, E., Nicolas, M. & Lenoir, J. (2021). The climatic debt is growing in the understorey of temperate forests: Stand characteristics matter. *Global Ecology and Biogeography*, 30 (7), 1474–1487. https://doi.org/10.1111/geb.13312
- Ritchie, A.J.C. (1955). Biological flora of the British Isles: Vaccinium vitis-idaea L . *Journal of Ecology*, 43 (2), 701–708. https://www.jstor.org/stable/2257030
- Ritchie, A.J.C. (1956). Biological flora of the British Isles: Vaccinium myrtillus L . Journal of Ecology, 44 (1), 291–299. https://www.jstor.org/stable/2257181
- Romeiro, J.M.N., Eid, T., Fernandez, C.A. & Kangas, A. (2022). Natural disturbances risks in European Boreal and Temperate forests and their links to climate change A review of modelling approaches. *Forest Ecology and Management*, 509, 120071. https://doi.org/10.1016/j.foreco.2022.120071
- Römermann, M., Gray, A., Vanbergen, A.J., Berges, L., Bohner, A., Brooker, R.W., Bryun, L. De, Cinti, B. De, Dirnböck, T., Grandin, U., Hester, A.J., Kanka, R., Klotz, S., Loucougaray, G., Lundin, L., Matteucci, G., Meszaros, I., Olah, V., Preda, E., Prevosto, B., Pykala, J., Schmidt, W., Taylor, M.E., Vadineanu, A., Waldmann, T. & Stadler, J. (2011). Functional traits and local environment predict vegetation responses to disturbance: a pan-European multi-site experiment. *Journal of Ecology*, 99 (3), 777–787. https://doi.org/https://doi.org/10.1111/j.1365-2745.2011.01794.x
- Roybal, C.M. & Butterfield, B.J. (2019). Species-specific trait–environment relationships among populations of widespread grass species. *Oecologia*, 189 (4), 1017–1026. https://doi.org/10.1007/s00442-019-04372-6
- Schlickmann, M.B., Carolina, A., Oliveira, L.M. De, Matteucci, D.O., Machado, F.D., Cuchi, T., Duarte, E. & Higuchi, P. (2020). Specific leaf area is a potential indicator of tree species sensitive to future climate change in the mixed

Subtropical Forests of southern Brazil. *Ecological Indicators*, 116, 106477. https://doi.org/10.1016/j.ecolind.2020.106477

- Seidl, R., Schelhaas, M., Rammer, W. & Verkerk, P.J. (2014). Increasing forest disturbances in Europe and their impact on carbon storage. *Nature Climate Change*, 4, 806–810. https://doi.org/10.1038/NCLIMATE2318
- Seidl, R., Thom, D., Kautz, M., Martin-Benito, D., Peltoniemi, M., Vacchiano, G., Wild, J., Ascoli, D., Petr, M., Honkaniemi, J. & Lexer, M.J. (2017). Forest disturbances under climate change. *Nature Publishing Group*, 7 (6), 395–402. https://doi.org/10.1038/nclimate3303
- Senf, C., Dirk, P., Zhiqiang, Y., Sebald, J., Knorn, J., Neumann, M. & Seidl, R. (2018). Canopy mortality has doubled in Europe's temperate forests over the last three decades. *Nature Communications*, 9, 4976. https://doi.org/10.1038/s41467-018-07539-6
- Sheffer, E., Canham, C.D. & Kigel, J. (2015). Countervailing effects on pine and oak leaf litter decomposition in human - altered Mediterranean ecosystems. *Oecologia*, 177, 1039–1051. https://doi.org/10.1007/s00442-015-3228-3
- Shome, A., Phartyal, S.S., Maharana, P. & Verma, A. (2023). Mapping peer reviewed scientific studies on plant trait – service linkages across ecosystems : A bibliometric analysis. *Anthropocene Science*, 2 (1), 19–30. https://doi.org/10.1007/s44177-023-00048-2
- Shovon, T.A., Rozendaal, D.M.A., Gagnon, D., Gendron, F., Vetter, M. & Vanderwel, M.C. (2019). Plant communities on nitrogen-rich soil are less sensitive to soil moisture than plant communities on nitrogen-poor soil. *Journal of Ecology*, 108 (1), 133–144. https://doi.org/https://doi.org/10.1111/1365-2745.13251
- Sjörs, H. (1999). The background: Geology, climate and zonation. In: Sjögren, E., Rydin, H., Snoeijs, P., & Diekmann, M. (eds) *Swedish plant geography*. Uppsala: Opilus Press AB. 5–15. http://www.divaportal.org/smash/get/diva2:565333/FULLTEXT01.pdf
- Sun, K., Sun, R., Li, Y., Ji, H., Jia, B. & Xu, Z. (2023). Plant economic strategies in two contrasting forests. *BMC Plant Biology*, 23, 366. https://bmcplantbiol.biomedcentral.com/articles/10.1186/s12870-023-04375-9
- Thom, D., Sommerfeld, A., Sebald, J., Hagge, J., Müller, J. & Seidl, R. (2020). Effects of disturbance patterns and deadwood on the microclimate in European beech forests. *Agricultural and Forest Meteorology*, 291, 108066. https://doi.org/10.1016/j.agrformet.2020.108066
- Vanneste, T., Valdés, A., Verheyen, K., Perring, M.P., Bernhardt-Römermann, M., Andrieu, E., Brunet, J., Cousins, S.A.O., Deconchat, M., De Smedt, P., Diekmann, M., Ehrmann, S., Heinken, T., Hermy, M., Kolb, A., Lenoir, J., Liira, J., Naaf, T., Paal, T., Wulf, M., Decocq, G. & De Frenne, P. (2019). Functional trait variation of forest understorey plant communities across Europe. *Basic and Applied Ecology*, 34, 1–14. https://doi.org/10.1016/j.baae.2018.09.004
- Villar, R. & Merino, J. (2001). Comparison of leaf construction costs in woody species with differing leaf life-spans in contrasting ecosystems. *New Phytologist*, 151 (1), 213–226. https://doi.org/10.1046/j.1469-8137.2001.00147.x

- Villar, R., Olmo, M., Atienza, P., Garzón, A.J., Wright, I.J., Poorter, H. & Hierro, L.A. (2021). Applying the economic concept of profitability to leaves. *Scientific Reports*, 11 (1), 1–10. https://doi.org/10.1038/s41598-020-79709-w
- Wasof, S., Lenoir, J., Gallet-Moron, E., Jamoneau, A., Brunet, J., Cousins, S.A.O., De Frenne, P., Diekmann, M., Hermy, M., Kolb, A., Liira, J., Verheyen, K., Wulf, M. & Decocq, G. (2013). Ecological niche shifts of understorey plants along a latitudinal gradient of temperate forests in north-western Europe. *Global Ecology and Biogeography*, 26 (10), 1138–1152. https://doi.org/10.1111/geb.12073
- Wickham, H., Chang, W. and Wickham, M.H. (2016). Package 'ggplot2'. *Create elegant* data visualisations using the grammar of graphics. Version, 2(1), pp.1-189.
- Woo, D.K. & Seo, Y. (2022). Effects of elevated temperature and abnormal precipitation on soil carbon and nitrogen dynamics in a Pinus densiflora forest. *Frontiers in Forests and Global Change*, 5, 1–15. https://doi.org/10.3389/ffgc.2022.1051210
- Wright, I.J., Reich, P.B. & Westoby, M. (2001). Strategy shifts in leaf physiology, structure and nutrient content between species of high- and low-rainfall and highand low-nutrient habitats. *Functional Ecology*, 15 (4), 423–434. https://doi.org/10.1046/j.0269-8463.2001.00542.x
- Xiao, R., Man, X., Duan, B., Cai, T., Ge, Z., Li, X. & Vesala, T. (2022). Changes in soil bacterial communities and nitrogen mineralization with understory vegetation in boreal larch forests. *Soil Biology and Biochemistry*, 166 (January), 108572. https://doi.org/10.1016/j.soilbio.2022.108572
- Yang, L., Wang, J., Geng, Y., Niu, S., Tian, D., Yan, T., Liu, W., Pan, J., Zhao, X. & Zhang, C. (2022). Heavy thinning reduces soil organic carbon: Evidence from a 9-year thinning experiment in a pine plantation. *Catena*, 211 (July 2021), 106013. https://doi.org/10.1016/j.catena.2021.106013
- Zellweger, F., De Frenne, P., Lenoir, J., Vangansbeke, P., Verheyen, K., Bernhardt-Römermann, M., Baeten, L., Hédl, R., Berki, I., Brunet, J., Van Calster, H., Chudomelová, M., Decocq, G., Dirnböck, T., Durak, T., Heinken, T., Jaroszewicz, B., Kopecký, M., Máliš, F., Macek, M., Malicki, M., Naaf, T., Nagel, T.A., Ortmann-Ajkai, A., Petřík, P., Pielech, R., Reczyńska, K., Schmidt, W., Standovár, T., Świerkosz, K., Teleki, B., Vild, O., Wulf, M. & Coomes, D. (2020). Forest microclimate dynamics drive plant responses to warming. *Science*, 368 (6496), 772–775. https://doi.org/10.1126/science.abf2939
- Zhang, S., Yang, X., Li, D., Li, S., Chen, Z. & Wu, J. (2022). A meta-analysis of understory plant removal impacts on soil properties in forest ecosystems. *Geoderma*, 426 (April), 116116. https://doi.org/10.1016/j.geoderma.2022.116116

# Popular science summary

#### Unravelling a mystery: What really influences Specific Leaf Area in Sweden's Understorey Plants

Have you ever wondered how plants decide to change their leaves' size? In the fascinating world of forest ecosystems, where overstorey trees stand tall and understorey plants thrive beneath the canopy, a captivating interaction takes place. Specific leaf area (SLA), the secret code to how plants grow and survive, is at the centre of this mystery. SLA is like a plant recipe that informs us how effectively a plant develops, adapts, and uses the resources around it. The change in SLA value offers signals on how plants could respond to changes in the environment, different light, and several nutrient conditions. Yet, this fascinating relationship between the trees and the understorey's leaves has remained hidden until now.

In this study, I explored the factors influencing the SLA of understorey plants across various forests in Sweden, including overstorey structure, soil characteristics, and the impact of climate induced temperature rise by developing a mathematical relationship between these variables and the SLA of understorey plants. To conduct this study, I used plant samples of two species, bilberry and cowberry, collected from forest stands which represent a forest that expands from South to North of Sweden, as well as from a mixed forest. Additionally, I used plant samples of wild strawberries and goldenrod from a unique experiment. This experiment involved selecting naturally growing plants from a cold region and replanting them into pots. Subsequently, these plants were relocated to a warmer region together with the same plant species of local origin. The experiment aimed to find if SLA of understorey plants would change by climate change induced temperature rise.

I found that in denser forests where trees were closer to each other, the SLA of understorey plants was higher. But having a lot of pine trees or broadleaf trees instead of Norway spruce around made the SLA of understorey plants decrease. Among the understorey plants, the newly grown leaves of cowberry had its unique way of growing when grown in dense forest of pine trees. Surprisingly, the characteristics of the soil did not matter to understorey plants. Furthermore, the SLA of wild strawberry and goldenrod did not vary across cold and local origins indicating these understorey plants of local origins changed their SLA in response to temperature increase. Additionally, SLA of wild strawberry and goldenrod decreased under denser trees even under the condition of climate induced temperature rise.

Such findings help us predict how changes in the forest due to climate change and forest management activities might affect the understorey plants. It can also help us protect biodiversity and manage the effects of climate change on forests similar to those studied here.

# Acknowledgements

As I reflect on the process of completing this thesis, my heart bursts with appreciation for everyone who has supported and guided me.

Special words of appreciation go to my family members for constantly supporting me throughout my life. I appreciate the motivation and constant support from my father, mother, and sister and all my other family members. It was your sacrifices and effort that has made me who I am today. My life journey has been guided by their enduring love, endless support, and unwavering trust in my abilities. I was unbroken through the challenging times by their prayers, and love. Further, I would not have made it without the love and support from Yojana, who has lifted me even in demotivating situations. I also want to thank Sugam for his assistance during the initial days.

My supervisors, Per-Ola Hedwall and Joan Díaz Calafat, have my utmost thanks. Thank you for the insightful guidance, steadfast support, and believing in my abilities to research on this topic. Thanks to their meticulous review and unwavering quest for excellence, my work has improved considerably. Further, I appreciate Joan Díaz Calafat and Laura Juvany Canovas for allowing me to use their data for this study. Without their efforts in data collection, this thesis would not be possible.

I am also very grateful to all professors at the University of Göttingen and Southern Swedish Forest Research Centre, Swedish University of Agriculture Sciences, who have constructed academic settings that have advanced my intellectual and personal development. Further, I am grateful for all resources available from the university.

A special thanks go to European Union for providing me Erasmus Mundus Joint Masters scholarship to complete my studies. I want to thank all my friends from the SUFONAMA cohort (2021-2023) and Euroforester cohort for the memorable time. A warm thanks to Hammed, Shakhera, Jeppe, Olamide, and Mary. Whether it be the time we spent travelling together or sitting in the classroom, it was all worth it.

### Publishing and archiving

Approved students' theses at SLU are published electronically. As a student, you have the copyright to your own work and need to approve the electronic publishing. If you check the box for **YES**, the full text (pdf file) and metadata will be visible and searchable online. If you check the box for **NO**, only the metadata and the abstract will be visible and searchable online. Nevertheless, when the document is uploaded it will still be archived as a digital file. If you are more than one author, the checked box will be applied to all authors. You will find a link to SLU's publishing agreement here:

• https://libanswers.slu.se/en/faq/228318.

 $\boxtimes$  YES, I/we hereby give permission to publish the present thesis in accordance with the SLU agreement regarding the transfer of the right to publish a work.

 $\Box$  NO, I/we do not give permission to publish the present work. The work will still be archived and its metadata and abstract will be visible and searchable.